UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

		Ŭ		
	Dat	te of report (Date of earliest event reported): November 29, 20	23	
		SKYE BIOSCIENCE, INC. (Exact name of registrant as specified in its charter)		
	Nevada	000-55136	45-0692882	
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification Number)	
		11250 El Camino Real, Suite 100, San Diego, CA 92130		
		(Address of principal executive offices)		
		(858) 410-0266		
		(Registrant's telephone number, including area code)		
		(Former name or former address, if changed since last report)		
Check	the appropriate box below if the Form 8-K filing	is intended to simultaneously satisfy the filing obligations of the	registrant under any of the following provisions.	
	Written communications pursuant to Rule 4	25 under the Securities Act (17 CFR 230.425)		
	Soliciting material pursuant to Rule 14a-12	under the Exchange Act (17 CFR 240.14a-12)		
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Securit	ies registered pursuant to Section 12(b) of the Ac	t:		
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
	N/A	N/A	N/A	

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).	e 12b-2 of
Emerging growth company []	
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box	financial

Item 1.01 Entry into a Material Definitive Agreement.

Tautomer Exclusive License Agreement

On November 30, 2023, Skye Bioscience, Inc., a Nevada corporation (the "Company") entered into an Exclusive License Agreement (the "License Agreement") with Tautomer Bioscience (Pty) Limited ("Tautomer"), pursuant to which, among other things, the Company granted to Tautomer an exclusive license to develop, manufacture and commercialize one or more products containing the Company's proprietary amino acid ester prodrug of delta-9-tetrahydrocannabinol (collectively, the "Products"), in the licensed field in the countries of the continent of Africa and their territories and possessions (the "Territory"). The licensed field includes (a) the prevention, treatment or control of humans of (i) chemotherapy induced nausea and vomiting or (ii) chronic pain of various etiologies and (b) the prevention, treatment or control in animals of any disease or medical condition, in each case (a) and (b), via suppository delivery.

Under the terms of the License Agreement, Tautomer will purchase a minimum of \$500,000 of shares of the Company's common stock within 60 days after the Effective Date (as defined in the License Agreement), in open market transactions complying with Rule 10b-18 of the Securities Exchange Act of 1934, as amended. The Company is entitled to receive from Tautomer milestone payments upon achievement of certain development, regulatory and commercial milestone events, for total potential milestone payments of up to \$10,750,000. In addition, Tautomer will pay to the Company tiered royalties up to the mid-double digits on the net sales of the Products in the Territory during the term of the License Agreement, subject to certain reductions for patent expiration and payments for licenses to third party patents. Tautomer will also pay to the Company a portion of all non-royalty sublicensing income received by Tautomer from any sublicensee.

Tautomer will be responsible for, at its own cost, and is required to use commercially reasonable efforts to, develop and commercialize the Products in the Territory, including all formulation, preclinical and clinical development and regulatory activities. Tautomer will reimburse the Company with markup for costs and expenses incurred by the Company for development activities. The Company will supply the Products to Tautomer during the term of the License Agreement.

The term of the License Agreement will continue on a market-by-market basis until expiration of the relevant royalty term of the Products, unless terminated earlier. Tautomer has the right to terminate License Agreement for any reason upon 90 days' written notice to the Company. The Company may terminate the License Agreement upon 90 days' written notice to Tautomer if (a) (i) the Company undergoes a Change of Control (as defined in the License Agreement) or (ii) the Company intends to grant an exclusive worldwide license for the licensed compound to a third party outside the licensed field and (b) the Company pays to Tautomer a tiered buy-out fee determined based on whether a commercial sale of the Products has occurred. In addition, either party may terminate the License Agreement for the other party's uncured breach, bankruptcy or insolvency or if Exchange Control Approval (as defined in the License Agreement) has not been obtained by April 28, 2024. The Product rights in the Territory will revert to the Company upon termination.

The foregoing description of the License Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the License Agreement, which is attached hereto as Exhibit 10.1 and incorporated herein by reference.

Sale of Avalite Lab Facility

On November 29, 2023, Avalite Sciences, Inc., a wholly-owned subsidiary of the Company ("Avalite") entered into an Offer to Sell (the "Offer to Sell") with Tab Labs Inc. (the "Purchaser") and Colliers Macaulay Nicolls Inc. (the "Agent"), pursuant to which Avalite agreed to sell to the Purchaser the Avalite property located at Unit 104 – 9295 198th Street, Langley, British Columba V1M 3J9, Canada (the "Property") for a purchase price of CAD \$1,499,000. The sale of the Property is subject to certain closing conditions, including the completion of the Purchaser's inspection of the Property and final approval of the transaction by the Purchaser's board of directors. The Company expects the closing of the sale of the Property to occur on January 15, 2024.

The foregoing description of the Offer to Sell does not purport to be complete and is qualified in its entirety by reference to the full text of the Offer to Sell, which is attached hereto as Exhibit 10.2 and incorporated herein by reference.

Item 1.02 Termination of a Material Definitive Agreement.

On December 1, 2023, the Company notified the University of Mississippi ("UM") that the Company was terminating, in its entirety, that certain Restated and Amended License Agreement, dated as of May 24, 2019, by and between the Company and UM (the "UM 8930 License Agreement"). In accordance with the terms of the UM 8930 License Agreement, the termination of such agreement will be effective on January 30, 2024 (the "Termination Date").

Pursuant to the UM 8930 License Agreement, UM granted the Company an exclusive license including, with the prior written consent of UM, the right to sublicense the intellectual property related to UM 8930 (referred to by Skye as SBI-200) for all fields of use.

The Company paid UM an upfront payment of \$200,000 under the UM 8930 License Agreement. Under the UM 8930 License Agreement, the Company was also responsible for an annual maintenance fee of \$75,000 that were credited against any royalties incurred. The aggregate milestone payments due by the Company under the UM 8930 License Agreement if all milestones were achieved was \$600,000 and the royalty percentage due on net sales was in the mid-single digits. The Company must also pay to UM a portion of all licensing fees received from any sublicensees, subject to a minimum royalty on net sales, and the Company is required to reimburse patent costs incurred by UM related to the licensed products. The royalty obligations applied by country and by licensed product, and ended upon the later of the date that no valid claim of a licensed patent covers a licensed product in a given country, or ten years after the first commercial sale of such licensed product in such country.

In March 2020, the Company paid UM a fee of \$200,000 as a result of notice of allowance being used by the United States Patent and Trademark Office for SBI-200, and as of the Termination Date, none of the other milestones under the UM 8930 License Agreement were met.

Effective as of the Termination Date, UM 8930 License Agreement will be terminated and will no longer be in effect, except that such termination does not relieve the parties from any obligation under the UM 8930 License Agreement that accrued prior to the termination or affect the survival of any other right, duty or obligation of the parties under the UM 8930 License Agreement, including certain other provisions expressly indicated to survive the termination.

The Company continues to hold an exclusive, perpetual license for UM 5050 from UM under that certain Restated and Amended License Agreement, dated as of May 24, 2019, by and between the Company and UM.

The foregoing description of the UM 8930 License Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the UM 8930 License Agreement, a copy of which is filed as Exhibit 10.13 to the Company's Annual Report on Form 10-K for the year ended December 31, 2022, filed with the SEC on March 31, 2023, and is incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On December 5, 2023, the Company issued a press release announcing the License Agreement. A copy of the press release is furnished as Exhibit 99.1 to this Current Report on Form 8-K

The information in Item 7.01 of this Current Report on Form 8-K, including the information in Exhibit 99.1 to this Current Report on Form 8-K, is furnished pursuant to Item 7.01 of Form 8-K and shall not be deemed "filed" for the purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section. Furthermore, the information in Item 7.01 of this Current Report on Form 8-K, including the information in Exhibit 99.1 to this Current Report on Form 8-K, shall not be deemed to be incorporated by reference in the filings of the Company under the Exchange Act or the Securities Act of 1933, as amended.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
10.1*	Exclusive License Agreement, dated November 30, 2023, by and between the Company and Tautomer Bioscience (Pty) Limited
10.2	Offer to Sell, dated November 29, 2023, by and among Colliers Macaulay Nicolls Inc., Tab Labs Inc. and Avalite Sciences, Inc.
99.1	Press release of Skye Bioscience, Inc. dated December 5, 2023
104	Cover Page Interactive Data File (embedded within the inline XBRL document)

^{*} Certain information has been omitted from this document in accordance with Regulation S-K, Item 601(b)(10).

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

SKYE BIOSCIENCE, INC.

Dated: December 5, 2023 /s/ Punit Dhillon

Name: Punit Dhillon

Title: Chief Executive Officer

Certain identified information has been excluded from the exhibit because it is both not material and is the type that the Registrant treats as private or confidential. Triple asterisks [***] denote exclusions.

EXCLUSIVE LICENSE AGREEMENT

This Exclusive License Agreement (the "Agreement") is entered into as of November 30, 2023 (the "Execution Date") by and between Skye Bioscience, Inc., a Nevada corporation, with an address of 11250 El Camino Real, Suite 100, San Diego, California, USA ("Skye"), and Tautomer Bioscience (Pty) Limited, a South African company with registered number 2016/450174/07 with an address of Woodmead North Office, 54 Maxwell Drive, Block B, Woodmead, 2191, Gauteng Province, South Africa ("Tautomer"). Skye and Tautomer are sometimes referred to herein individually as a "Party" and collectively as the "Parties".

RECITALS

Whereas, Skye has developed an amino acid ester prodrug of delta-9-tetrahydrocannabinol and possesses certain intellectual property related to such prodrug;

Whereas, Tautomer is a biopharmaceutical company focused on development of therapeutic products; and

Whereas, Skye desires to grant Tautomer an exclusive license, and Tautomer desires to obtain a license, to Develop, Manufacture and Commercialize Licensed Products in the Field in the Territory (as each such term is defined below).

Now, Therefore, in consideration of the foregoing premises and the mutual promises, covenants and conditions contained in this Agreement, the Parties agree as follows:

Article 1 DEFINITIONS

- 1.1 "Accounting Standards" means either GAAP or IFRS, as designated and used by the applicable Party in preparing its financial statements from time to time.
- 1.2 "Affiliate" means, with respect to a particular Party, a person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such Party. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, or by contract or otherwise.
 - **1.3** "Anti-Corruption Laws" has the meaning set forth in Section 10.6(a).
- 1.4 "Business Day" means a day other than Saturday, Sunday or any day that banks in San Diego, California, United States, South Africa are required or permitted to be closed.
 - 1.5 "Buy-Out Fee" has the meaning set forth in Section 13.3.

- 1.6 "Calendar Quarter" means the four quarters of a Calendar Year, each Calendar Quarter starting on January 1, April 1, July 1 and October 1, except that the first Calendar Quarter of the Term shall commence on the Effective Date and the last Calendar Quarter of the Term shall end on the last day of the Term.
- 1.7 "Calendar Year" means the period beginning on January 1 and ending on December 31, except for the first Calendar Year of the Term that shall begin on the Effective Date and end on December 31 of the year during which the Effective Date occurs, and the last Calendar Year of the Term shall commence on January 1 of the year in which the Term ends and end on the last day of the Terms.
- 1.8 "Change of Control" means, with respect to a Party: (a) the sale of all or substantially all of its assets or all of its assets relating to this Agreement; (b) a merger, reorganization or consolidation involving such Party in which the holders of the voting securities of such Party outstanding immediately prior thereto cease to beneficially own at least fifty percent (50%) of the combined voting power of the surviving entity, directly or indirectly, immediately after such merger, reorganization or consolidation; or (c) a transaction in which an entity or individual, or group of entities and/or individuals acting in concert, acquires more than fifty percent (50%) of the voting equity securities of such Party.
- 1.9 "Chemotherapy Induced Nausea and Vomiting" or "CINV" means nausea and/or vomiting as a complication of chemotherapy. CINV can be classified into 5 sub-types: acute, delayed, anticipatory, breakthrough and refractory. Acute CINV occurs within 24 hours of the initial administration of an antineoplastic agent, while delayed CINV occurs after 24 hours and may peak 2 to 3 days post administration. Once a patient experiences CINV, he or she may then experience anticipatory CINV, which occurs when a sensory experience (e.g., smell, sound, taste) triggers an episode of nausea and/or vomiting prior to subsequent administration of a chemotherapy regimen. Breakthrough CINV can be defined as nausea and/or vomiting that occurs within 5 days of chemotherapy treatment despite the use of a guideline-recommended antiemetic protocol, which requires the addition of more agents referred to as "rescue medications." Refractory CINV can be described as nausea and/or vomiting that consistently occurs in subsequent chemotherapy cycles despite the use of a guideline-recommended antiemetic regimen.
- 1.10 "Chronic Pain" means any pain that lasts for over three (3) consecutive months. Chronic pain may include back pain, cancer pain, pelvic pain and neuropathic pain.
 - **1.11** "Claims" has the meaning set forth in Section 11.1.
- 1.12 "COGS" stands for cost of goods sold and means the documented, fully allocated internal and external costs of Manufacturing, raw materials, reservation or similar capacity fees, packaging (including bulk and temporary packaging, pallets, palletizing etc.), labor, capital expenditures, forecasting, ordering, inventory management, quality control, stability testing, release costs, warehousing, transportation, import/export costs and other costs ordinarily included as a cost of goods sold under Accounting Standards. Where Skye procures supplies or any step or part thereof from a Contract Manufacturer or other contractor, the price paid by Skye to the Contract Manufacturer, or such contractor shall be used when determining the applicable external COGS or portion thereof.

- 1.13 "Combination Product" means: (a) a pharmaceutical product that consists of a Licensed Product and at least one other clinically active ingredient that is not a Licensed Product; or (b) any combination of a Licensed Product and another pharmaceutical product that contains at least one other clinically active ingredient that is not a Licensed Product, where such products are not formulated together but are sold together as a single product and invoiced as one product. The other clinically active ingredient(s) in clause (a) and the other pharmaceutical product(s) in clause (b) are each referred to as the "Other Product(s)".
- 1.14 "Commercialization" or "Commercialize" means any and all activities directed to the offering for sale and sale of Licensed Product, including (a) marketing, promoting, advertising, exhibiting, distributing, detailing, selling (and offering for sale or contracting to sell) or otherwise commercially exploiting a Licensed Product in the Field in the Territory (including importing and exporting activities in connection therewith); (b) order processing, handling of returns and recalls, booking of sales and transporting such Licensed Product for commercial sale; (c) the conduct of any post-approval clinical trials involving such Licensed Product; (d) interacting with Regulatory Authorities regarding the above; and (e) seeking and obtaining pricing approvals and reimbursement approvals (as applicable) for that Licensed Product in the Territory. For clarity, Commercialization does not include manufacture of Licensed Product.
 - **1.15** "Commercialization Plan" has the meaning set forth in Section 7.3.
- 1.16 "Commercially Reasonable Efforts" means, with respect to the efforts and resources to be expended, or considerations to be undertaken by Tautomer with respect to any objective, activity, or decision to be undertaken hereunder with respect to the Development or Commercialization of a Licensed Product, the level of efforts and allocation of resources that a similarly-situated pharmaceutical company would reasonably devote to carry out such activities for a product of similar market potential and at a similar stage of product life in a diligent and sustained manner without undue interruption or delay.
 - 1.17 "Competing Product" has the meaning set forth in Section 3.4(a).
 - **1.18** "Competing Program" has the meaning set forth in Section 3.4(b).
- **1.19** "Compound IP" means any and all Know-How, and intellectual property rights therein (including Patents Covering such Know-How), that is discovered, made, conceived or first reduced to practice under this Agreement and relates to the Licensed Compound.
- 1.20 "Confidential Information" of a Party means any and all information that is disclosed by or on behalf of one Party or its Affiliates ("Disclosing Party") to the other Party or its Affiliates ("Receiving Party") pursuant to this Agreement (including information disclosed prior to the Execution Date pursuant to the Confidentiality Agreement), whether in oral, written, graphic or electronic form. The terms of this Agreement shall be deemed the Confidential Information of both Parties. All Know-How that is discovered, made, generated conceived or reduced to practice under this Agreement shall be deemed the Confidential Information of the Party that owns such Know-How, who shall be deemed the Disclosing Party with respect thereto and the other Party shall be deemed the Receiving Party.

- **1.21** "Confidentiality Agreement" means that Confidentiality Agreement between Skye and Tautomer dated as of November 2, 2022.
- 1.22 "Control" means, with respect to any material, Know-How, or intellectual property right, that a Party (a) owns or (b) has a license (other than a license granted to such Party under this Agreement) to such material, Know-How, or intellectual property right and, in each case, has the ability to grant to the other Party access, a license, or a sublicense (as applicable) to the foregoing on the terms and conditions set forth in this Agreement without violating the terms of any then-existing agreement or other legally enforceable arrangement with any Third Party. Notwithstanding the foregoing, a Party will not be deemed to "Control" any material, Know-How or intellectual property right (including Patents) that, prior to the consummation of Change of Control that makes a Third Party an Affiliate of such Party, is owned or controlled by such Third Party (except for any material, Know-How or intellectual property right that was licensed to such Party prior to such Change of Control). With respect to any material, Know-How or intellectual property right (including Patents) obtained by Skye or its Affiliates from a Third Party after the Execution Date, Skye or its Affiliates, as applicable, shall be deemed to Control such material, Know-How or intellectual property right only if it possesses the right to grant such license, sublicense, access or rights to use without being obligated to pay any royalties or other consideration therefor (excluding any payment obligations with respect to Skye's or its Affiliates' employees or contractors), unless Tautomer (at its option) agrees in advance of any grant of rights thereto to pay all royalties and other consideration owed to such Third Party arising specifically as a result of Tautomer's or any of its Affiliates's or Sublicensee's use or practice of such material, Know-How or intellectual property right.
 - 1.23 "Contract Manufacturer" means any contract manufacturer engaged by Skye to Manufacture and supply Licensed Compound.
- **1.24** "Cover" means, with respect to a claim of a Patent and a Licensed Product, that such claim would be infringed, absent a license, by the use, manufacture, offer for sale, sale or importation of such Licensed Product (considering claims of patent applications to be issued as then pending).
- 1.25 "Develop" or "Development" means all activities that relate to the development of Licensed Products or to obtaining, maintaining or expanding Regulatory Approval of a Licensed Products, including, but not limited to (a) formulation development of Licensed Products; (b) preclinical testing, toxicology and clinical trials; (c) preparation, submission, review and development of data or information for the purpose of submission to a Governmental Authority to obtain, maintain or expand Regulatory Approval of a Licensed Product; and (d) manufacturing process development for preclinical testing and clinical trials, and related quality assurance and technical support activities. "Develop" has a correlative meaning.
 - **1.26** "Development Plan" has the meaning set forth in Section 5.2.
 - **1.27** "Disclosing Party" has the meaning set forth in Section 1.20.
 - **1.28** "**Dispute**" has the meaning set forth in Section 14.1.
 - **1.29** "Divestiture" has the meaning set forth in Section 3.4(b)(ii).

- **1.30** "Dollar" means a U.S. dollar, and "\$" shall be interpreted accordingly.
- 1.31 "Effective Date" means the date on which the Suspensive Condition is fulfilled pursuant to Section 2.1
- **1.32** "Exchange Act" has the meaning set forth in Section 8.1.
- 1.33 "Exchange Control Approval" means the approval by the Financial Surveillance Department of the South African Reserve Bank, as may be required in terms of the Exchange Control Regulations (issued in terms of the Currency and Exchanges Act 9 of 1933) for any of the transactions contemplated in this Agreement.
 - 1.34 "Exchange Control Approval Period" has the meaning set forth in Section 2.1.
- **1.35** "Executive Officer" means, with respect to Skye, its Chief Executive Officer, and with respect to Tautomer, its Chief Executive Officer, or, in either case, a designee with senior decision-making authority.
- 1.36 "Export Control Laws" means: (a) all applicable U.S. laws and regulations relating to sanctions and embargoes imposed by U.S. Department of Treasury's Office of Foreign Assets Control (or its successor office or other body having substantially the same function); (b) all applicable U.S. export control laws, including the Arms Export Controls Act (22 U.S.C. Ch. 39), the International Emergency Economic Powers Act (50 U.S.C. §§ 1701 et seq.), the Trading With the Enemy Act (50 U.S.C. app. §§ 1 et seq.), the Export Administration Act of 1979 (50 U.S.C. app. §§ 2401 et seq.), International Boycott Provisions of Section 999 of the U.S. Internal Revenue Code of 1986 and all rules, regulations and executive orders relating to any of the foregoing, including but not limited to the International Traffic in Arms Regulations (22 C.F.R. §§ 120 et seq.), the Export Administration Regulations (15 C.F.R. §§ 730 et seq.) and the regulations administered by the Office of Foreign Assets Controls of the United States Department of the Treasury; and (c) all export controls imposed on any Licensed Compound or Licensed Product by any country or organization or nation within the jurisdiction of which either Party operates or does business.
 - 1.37 "FD&C Act" means the U.S. Federal Food, Drug and Cosmetic Act, as amended.
 - **1.38** "FDA" means the U.S. Food and Drug Administration or any successor entity.
- 1.39 "Field" means (a) the prevention, treatment or control in humans of (i) Chemotherapy Induced Nausea and Vomiting or (ii) Chronic Pain of various etiologies, including neuropathic pain and (b) the prevention, treatment or control in animals of any disease or medical condition, in each case (a) and (b), via suppository delivery. For avoidance of doubt, the Field does not include any diseases or medical conditions in humans other than Chemotherapy Induced Nausea and Vomiting and Chronic Pain.
- 1.40 "First Commercial Sale" means, with respect to a Licensed Product, the first sale on a commercial basis to a Third Party of such Licensed Product in a given regulatory jurisdiction after Regulatory Approval has been obtained in such jurisdiction for such Licensed Product.

- 1.41 "GAAP" means generally accepted accounting principles in the U.S., consistently applied.
- 1.42 "GCP" means all applicable good clinical practice standards for the design, conduct, performance, monitoring, auditing, recording, analyses and reporting of clinical trials, including, as applicable (a) as set forth in the ICH Harmonized Tripartite Guideline for Good Clinical Practice (CPMP/ICH/135/95) (the "ICH Guidelines") and any other guidelines for good clinical practice for trials on medicinal products in the Territory and (b) the equivalent applicable Law in each country in the Territory, each as may be amended and applicable from time to time and in each case, that provide for, among other things, assurance that the clinical data and reported results are credible and accurate and protect the rights, integrity and confidentiality of trial subjects.
- 1.43 "GLP" means current good laboratory practices as established by any Regulatory Authority in the Territory and as interpreted by relevant ICH Guidelines; in each case, as amended from time to time.
- 1.44 "Governmental Authority" means any multi-national, federal, state, local, municipal, provincial or other governmental authority of any nature (including any governmental division, prefecture, subdivision, department, agency, bureau, branch, office, commission, council, court or other tribunal).
- 1.45 "ICH" means the International Conference on Harmonization of Technical Requirements for Registration of Pharmaceuticals for Human Use.
 - **1.46** "**ICH Guidelines**" has the meaning set forth in Section 1.41.
 - 1.47 "IFRS" means international financial reporting standards, consistently applied.
- **1.48** "IND" means (a) an Investigational New Drug Application as defined in the FD&C Act and applicable regulations promulgated thereunder by the FDA or (b) the equivalent application to the equivalent agency in any other regulatory jurisdiction, the filing of which is necessary to initiate or conduct clinical testing of a pharmaceutical product in humans in such jurisdiction.
 - **1.49** "Indemnified Party" has the meaning set forth in Section 11.2.
 - **1.50** "Indemnifying Party" has the meaning set forth in Section 11.2.
 - **1.51** "Indication" means Chemotherapy Induced Nausea and Vomiting or Chronic Pain.
- **1.52** "Indirect Taxes" means any and all sales, use, excise, export or import, consumption, goods and services, withholding, value added or other similar taxes, government permit or license fees and any and all customs, duty, tariff and other similar fees levied upon the transactions contemplated by this Agreement.
 - **1.53** "**Insolvency Event**" has the meaning set forth in Section 13.5.

- **1.54** "**Joint IP**" has the meaning set forth in Section 9.1.
- **1.55** "**Joint Patents**" has the meaning set forth in Section 9.1.
- **1.56** "Joint Steering Committee" or "JSC" has the meaning set forth in Section 4.1.
- **1.57** "Know-How" means any and all data (including clinical data), results, technology or information, including know-how, trade secrets, practices, techniques, methods, processes, inventions, developments, specifications, formulations, formulae, software, algorithms, marketing reports, expertise, technology, test data (including pharmacological, biological and chemical, biochemical, clinical test data and data resulting from non-clinical studies) and other study data and procedures.
- **1.58** "Laws" means all laws, statutes, rules, regulations ordinances and other pronouncements having the effect of law of any federal, national, multinational, state, provincial, county, city or other political subdivision, domestic or foreign.
- **1.59** "Licensed Compound" means Skye's proprietary amino acid ester prodrug of delta-9-tetrahydrocannabinol, internally coded by Skye as "SBI-100" and also known as "THCVHS", as further described in <u>Schedule 1.58</u>.
- **1.60** "Licensed Know-How" means all Know-How Controlled by Skye or its Affiliates as of the Execution Date, between the Execution Date and the Effective Date or during the Term that is necessary or useful to Develop, Manufacture or Commercialize Licensed Products, but excluding Know-How that is necessary or useful to Manufacture Licensed Compound.
- 1.61 "Licensed Patents" means the Patents Controlled by Skye or its Affiliates as of the Execution Date, between the Execution Date and the Effective Date or during the Term that are necessary to Develop, Manufacture or Commercialize Licensed Products, but excluding Patents that are necessary to Manufacture Licensed Compound.
- **1.62** "Licensed Product" means any pharmaceutical product, formulated as a suppository, containing a Licensed Compound as any ingredient.
- 1.63 "Licensed Technology" means the Licensed Patents, Licensed Know-How, Licensed Trademarks and Skye's rights under the Joint IP.
- **1.64** "Licensed Trademarks" means trademarks Controlled by Skye or its Affiliates as of the Execution Date, between the Execution Date and the Effective Date or during the Term that are selected by Skye to brand or label Licensed Products in the Field in the Territory.
- **1.65** "Manufacture" and "Manufacturing" means all activities related to the making, production, processing, filling, finishing, testing, packaging, labelling, shipping and holding of Licensed Compound and Licensed Products.

- **1.66** "Marketing Authorization Application" or "MAA" means an application to the appropriate Regulatory Authority for approval to market a Licensed Product (but excluding Pricing Approval) in any particular jurisdiction.
- **1.67** "Net Sales" means, with respect to any Licensed Product, the gross amounts invoiced by Tautomer and its Affiliates and Sublicensees for sales of such Licensed Product to Third Parties, less the following deductions to the extent actually allowed and taken:
- (a) cash, trade or quantity discounts, charge-back payments and rebates actually granted to trade customers, retail pharmacy chains, wholesalers, managed health care organizations, pharmaceutical benefit managers, insurers, group purchasing organizations and national, state or local government;
- (b) credits, rebates or allowances actually allowed upon prompt payment or on account of claims, damaged goods, rejections or returns of Licensed Products, including in connection with recalls, and the actual amount of any write-offs for bad debt, such write-offs not to exceed two percent (2%) of Net Sales (provided that any amount subsequently recovered will be treated as Net Sales);
 - (c) freight, postage, shipping, transportation and insurance charges, in each case actually allowed or paid; and
- (d) taxes (other than income taxes), duties, tariffs, mandated contributions or other governmental charges levied on the sale of Licensed Products to Third Parties, including Indirect Taxes and excise taxes on such sales, to the extent separately in the invoice and actually paid by the selling party.

Net Sales shall be determined in accordance with the Accounting Standards. If a single item falls into more than one of the categories set forth in clauses (a)-(d) above, such item may not be deducted more than once.

Sales among Tautomer and its Affiliates or Sublicensees, which are subsequently resold or to be resold by Tautomer, its Affiliate or Sublicensee will not be deemed a sale within the meaning of this definition, but in such cases Net Sales will accrue and be calculated on any subsequent sale or other transfer to a person who is not an Affiliate of Tautomer or a Sublicensee.

Net Sales shall not include any amounts invoiced for sales of Licensed Products supplied for use in clinical trials of Licensed Products, or under early access, compassionate use or named patient programs; provided that, in each case, such sales are at or below seller's costs.

Net Sales for a Combination Product in a country shall be calculated as follows:

(i) If the Licensed Product and Other Product(s) each are sold separately in such country, Net Sales will be calculated by multiplying the total Net Sales (as described above) of the Combination Product by the fraction A/(A+B), where A is the public or list price in such country of the Licensed Product sold separately in the same formulation and dosage, and B is the

(sum of the) public or list price(s) in such country of the Other Product(s) sold separately in the same formulation and dosage, during the applicable Calendar Year.

- (ii) If the Licensed Product is sold independently of the Other Product(s) in such country, but the public or list price of the Other Product(s) cannot be determined, Net Sales will be calculated by multiplying the total Net Sales (as described above) of such Combination Product by the fraction A/C, where A is the public or list price in such country of such Licensed Product sold independently and C is the public or list price in such country of the Combination Product.
- (iii) If the Other Product(s) are sold independently of the Licensed Product therein in such country, but the public or list price of such Licensed Product cannot be determined, Net Sales will be calculated by multiplying the total Net Sales (as described above) of such Combination Product by the fraction 1-B/C, where B is the (sum of the) public or list price(s) in such country of the Other Product(s) and C is the public or list price in such country of the Combination Product.
- (iv) If neither the Licensed Product nor the Other Product(s) is sold independently in such country a market price for the Licensed Product and the Other Product(s) shall be negotiated by the Parties in good faith based upon the costs, overhead and profit as are then incurred for such Combination Product.
- "Non-Royalty Sublicensing Income" means all amounts received by Tautomer and its Affiliates from any Sublicensee as consideration for the grant of intellectual property rights under an agreement with a Third Party that includes the grant to such Sublicensee of a sublicense under, an option or a covenant not to sue with respect to any Licensed Technology. Without limiting the generality of the foregoing, Non-Royalty Sublicensing Income shall include up-front fees, license fees, milestone payments, technology access fees, license maintenance fees, premiums above the fair market value on sales of equity or convertible debt securities of Tautomer or its Affiliate (as determined as set forth below in this paragraph) and any other payments attributable to the grant to such Sublicensee of a license or sublicense of any intellectual property rights. Notwithstanding the foregoing, Non-Royalty Sublicensing Income shall exclude: (a) royalties or profit share payments paid or payable by any Sublicensee to Tautomer or its Affiliates with respect to sales or other dispositions of Licensed Products by Sublicensees; (b) bona fide Development funding paid by a Sublicensee to Tautomer or its Affiliates to reimburse expenses incurred by Tautomer or its Affiliates specifically in the performance of Development activities with respect to Licensed Products conducted after the date of the sublicense; and (c) payments for equity or convertible debt securities of Tautomer or its Affiliates that are at or below the fair market value of such securities on the date of receipt, either (i) as reasonably determined in good faith by Tautomer's or its Affiliate's board of directors or governing body of Tautomer or its Affiliates having the right to make such determination, if such securities are not then traded on a public securities exchange or (ii) as determined by the closing price of such securities of Tautomer or its Affiliates (as applicable) on the date of receipt (or the most recent trading day on such exchange preceding such receipt), if such securities are then traded on a public securities exchange, provided that, in either case, Tautomer shall provide reasonably detailed, true and accurate supporting documentation with respect to such fair market value.

- **1.69** "Other Party" has the meaning set forth in Section 9.8.
- 1.70 "Other Product(s)" has the meaning set forth in Section 1.13.
- 1.71 "Patents" means (a) pending patent applications, issued patents, utility models and designs; (b) reissues, substitutions, confirmations, registrations, validations, re-examinations, additions, continuations, continued prosecution applications, continuations-in-part or divisions of or to any of the foregoing; and (c) extensions, renewals or restorations of any of the foregoing by existing or future extension, renewal or restoration mechanisms, including supplementary protection certificates or the equivalent thereof.
 - **1.72** "Pharmacovigilance Agreement" has the meaning set forth in Section 5.11.
- 1.73 "Pricing Approval" means such governmental approval, agreement, determination or decision establishing prices for a Licensed Product that can be charged and/or reimbursed in regulatory jurisdictions where the applicable Governmental Authorities approve or determine the price and/or reimbursement of pharmaceutical products.
 - **1.74** "**Product Infringement**" has the meaning set forth in Section 9.4(a).
 - **1.75** "**Publication**" has the meaning set forth in Section 12.5.
 - **1.76** "Receiving Party" has the meaning set forth in Section 1.20.
- 1.77 "Regulatory Approval" means all approvals that are necessary for the commercial sale of a Licensed Product in the Field in a given country or regulatory jurisdiction.
- **1.78** "Regulatory Authority" means, in a particular country or jurisdiction, any applicable Governmental Authority involved in granting Regulatory Approval in such country or jurisdiction.
- 1.79 "Regulatory Materials" means regulatory applications, submissions, notifications, communications, correspondence, registrations, Regulatory Approvals and/or other filings made to, received from or otherwise conducted with a Regulatory Authority in order to Develop, Manufacture, market, sell or otherwise Commercialize a Licensed Product in a particular country or jurisdiction.
 - **1.80** "**Rejecting Party**" has the meaning set forth in Section 3.6.
 - **1.81** "Requesting Party" has the meaning set forth in Section 9.8.
 - **1.82** "Review Period" has the meaning set forth in Section 12.5.
 - **1.83** "Royalty Term" has the meaning set forth in Section 8.5(b).
 - **1.84** "Rule 10 b-18" has the meaning set forth in Section 8.1.
 - **1.85** "Rules" has the meaning set forth in Section 14.2(a).

- **1.86** "Senior Executives" means (a) in the case of Skye, the Chief Executive Officer of Skye (or a senior executive officer designated by the Chief Executive Officer) and (b) in the case of Tautomer, [the Chief Executive Officer of Tautomer (or a senior executive officer designated by the Chief Executive Officer)].
 - **1.87** "Skye Development Expenses" has the meaning set forth in Section 8.2.
 - **1.88** "Skye Indemnitees" has the meaning set forth in Section 11.1.
 - **1.89** "Sole Inventions" has the meaning set forth in Section 9.1.
- **1.90** "Sublicensee" means any Third Party granted a sublicense, option or covenant not to sue by Tautomer to the rights licensed to Tautomer under Section 3.1(a).
 - **1.91** "Supply Agreement" has the meaning set forth in Section 6.3.
 - **1.92** "Suspensive Condition" has the meaning set forth in Section 2.1.
 - **1.93** "**Tautomer Indemnitees**" has the meaning set forth in Section 11.1.
- 1.94 "Tautomer IP" means any and all Know-How, and intellectual property rights therein and thereto (including patents and patent applications covering such Know-How), owned or controlled by Tautomer as of the Execution Date, between the Execution Date and the Effective Date or during the Term that is necessary or useful to Develop, Manufacture, Commercialize or otherwise exploit Licensed Compounds or Licensed Products, including any Know-How that is discovered, made, conceived or first reduced to practice in connection with the research, Development, Manufacture or Commercialization of a Licensed Compound or Licensed Product, excluding Compound IP and Joint IP.
 - **1.95** "**Tautomer Patents**" has the meaning set forth in Section 9.3(a).
 - **1.96** "Term" has the meaning set forth in Section 13.1.
 - **1.97** "**Territory**" means the countries of the continent of Africa and their territories and possessions.
 - **1.98** "Third Party" means any entity other than Skye or Tautomer or an Affiliate of either of them.
 - 1.99 "U.S." means the United States of America, including all possessions and territories thereof.
- 1.100 "Valid Claim" means (a) a claim of an issued, unexpired patent within the Licensed Patents that has not been revoked, disclaimed, abandoned or held invalid or unenforceable by a court or other body of competent jurisdiction in an unappealed or unappealable decision and (b) a claim of any patent application within the Licensed Patents which has not been abandoned or finally disallowed without the possibility of appeal or re-filing of the application.

Article 2 SUSPENSIVE CONDITION

- **2.1 Exchange Control Approval**. The provisions of this Agreement (except as set forth in Section 13.1) are subject to, and conditional upon, the fulfilment of the Suspensive Condition on or before one hundred fifty (150) days from the Execution Date (or such later date and/or time as may be agreed on by the Parties) (the "**Exchange Control Approval Period**"). "**Suspensive Condition**" means the requirement to obtain Exchange Control Approval without conditions (or if the Exchange Control Approval is granted conditionally, this Suspensive Condition shall only be regarded as having been fulfilled if Skye has accepted the conditions in writing).
- **2.2 Efforts**. Each of the Parties shall use its commercially reasonable efforts and shall take all such steps and render all such assistance to the other Party as may be reasonably necessary to procure fulfilment of the Suspensive Condition during the Exchange Control Approval Period.

Article 3 LICENSES AND EXCLUSIVITY

3.1 License Grant to Tautomer.

- (a) License to Tautomer. Subject to the terms and conditions of this Agreement, Skye hereby grants to Tautomer an exclusive, royalty-bearing license, with the right to sublicense through multiple tiers in accordance with Section 3.1(b), under the Licensed Technology (a) to Develop, Commercialize, register, sell and offer for sale Licensed Products in the Field in the Territory and (b) to Manufacture Licensed Products (but not Licensed Compound) in South Africa, solely from Licensed Compound supplied by or behalf of Skye, for Development, Commercialization and sale of Licensed Products in the Field in the Territory.
- (b) Sublicenses. Tautomer may grant sublicenses through multiple tiers, under any or all of the rights granted in Section 3.1(a), to its Affiliates and to Third Parties. Any sublicenses granted to a Third Party, other than sublicenses granted to Third Party subcontractors performing services on behalf of Tautomer, shall require the prior written consent of Skye, such consent not to be unreasonably withheld, conditioned or delayed. Each sublicense shall be subject to a written agreement that is consistent with the terms and conditions of this Agreement, and Tautomer shall ensure that its Sublicensees comply with the terms and conditions of such agreement. Notwithstanding any such sublicense, Tautomer shall remain solely liable for the performance of its obligations hereunder. Tautomer shall promptly notify Skye in writing of a sublicense and provide to Skye a copy of such sublicense within thirty (30) days after execution; provided that Tautomer may redact any confidential or financial information contained therein that is unnecessary for Skye to confirm compliance with this Agreement.
- 3.2 Skye Retained Rights. Notwithstanding the exclusive nature of the license granted to Tautomer pursuant to Section 3.1(a), Skye expressly retains rights under the Licensed Technology (a) to perform its obligations under this Agreement and the Supply Agreement and

(b) to Develop and Manufacture Licensed Compound in the Territory to support Development, Commercialization and any other exploitation of Licensed Compound outside of the Territory.

3.3 License and Option Grant to Skye.

- (a) Tautomer hereby grants to Skye a royalty-free, sublicensable (through multiple tiers), perpetual, irrevocable, non-exclusive license to practice the Tautomer IP to perform its obligations under this Agreement and the Supply Agreement.
- (b) Tautomer hereby grants to Skye an exclusive option to obtain (i) an exclusive license under the Tautomer IP and Tautomer's rights under the Joint IP to Develop, Manufacture, Commercialize and otherwise exploit Licensed Compound and Licensed Products outside of the Territory and (ii) a non-exclusive license to practice the Tautomer IP to Develop and Manufacture Licensed Compound and Licensed Products in the Territory to support Development, Commercialization and any other exploitation of Licensed Compound and Licensed Products outside of the Territory. Skye may exercise such option at any time during the Term by providing written notice to Tautomer. Promptly after Tautomer's receipt of such notice, the Parties shall negotiate in good faith a license agreement to grant Skye the rights set forth in this Section 3.3(b), such license agreement to include customary and commercially reasonable terms consistent with similar licensing arrangements and other mutually agreeable terms and conditions. The Parties shall use commercially reasonable efforts to execute such license agreement no later than four (4) months after Tautomer's receipt of such notice.

3.4 Exclusivity.

- (a) Competing Product. During the Term, Tautomer and its Affiliates shall not, by itself or by granting, directly or indirectly, to any Third Party the right to, research, Develop, Manufacture or commercialize any pharmaceutical product formulated as a suppository that contains delta-9-tetrahydrocannabinol (a "Competing Product") in the Territory, other than Licensed Products.
- **(b)** Acquisition of Competing Program. In the event that a Third Party becomes an Affiliate of Tautomer after the Effective Date through merger, acquisition, consolidation or other similar transaction, and, as of the closing date of such transaction, such Third Party is engaged in the Development, Manufacture or Commercialization of a pharmaceutical product that, if conducted by such new Affiliate, would cause Tautomer to be in breach of its exclusivity obligations set forth above (a "Competing Program"), then:
- (i) if such transaction results in a Change of Control of Tautomer, then such new Affiliate may continue such Competing Program and such continuation will not constitute a breach of Tautomer's exclusivity obligations in Section 3.3(b)(a); provided that (A) such new Affiliate conducts such Competing Program independently of the activities of this Agreement, (B) such new Affiliate does not use or access any of the Licensed Technology or Confidential Information of Skye or its Affiliates in the conduct of such Competing Program and (C) no personnel involved in performing, overseeing, monitoring or making decisions with respect to the Competing Program have access to non-public plans or information relating to the Development, Manufacture or Commercialization of Licensed Products, excluding executive

management personnel reviewing and evaluating plans and information in connection with portfolio decision-making among product opportunities; and

- (ii) if such transaction does not result in a Change of Control of Tautomer, then Tautomer and its new Affiliate will have twelve (12) months from the closing date of such transaction to wind down or complete the Divestiture of such Competing Program, and the new Affiliate's conduct of such Competing Program during such twelve (12)-month period will not be deemed a breach of Tautomer's exclusivity obligations in Section 3.3(b)(a); provided that (A) such new Affiliate conducts such Competing Program independently of the activities of this Agreement, (B) such new Affiliate does not use or access any of the Licensed Technology or Confidential Information of Skye or its Affiliates in the conduct of such Competing Program and (C) no personnel involved in performing, overseeing, monitoring or making decisions with respect to the Competing Program have access to non-public plans or information relating to the Development, Manufacture or Commercialization of Licensed Products, excluding executive management personnel reviewing and evaluating plans and information in connection with portfolio decision-making among product opportunities. "Divestiture" means the sale or transfer of rights to the Competing Program to a Third Party without receiving a continuing share of profit, royalty payment or other economic interest in the success of such Competing Program.
- 3.5 No Implied Licenses; Negative Covenant. Except as explicitly set forth in this Agreement, neither Party shall be deemed by estoppel or implication to have granted the other Party any license or other right to any intellectual property of such Party. Tautomer shall not, and shall not permit any of its Affiliates or Sublicensees to, practice any Licensed Technology outside the scope of the license granted to it under this Agreement.
- 3.6 Insolvency. If this Agreement is terminated due to the rejection of this Agreement by or on behalf of either Party due to an Insolvency Event of such Party (the "Rejecting Party"), all licenses and rights to licenses granted under or pursuant to this Agreement by the Rejecting Party to the other Party are and shall otherwise be deemed to be licenses of rights to "intellectual property" (including for purposes of Section 365(n) of Title 11 of the United States Bankruptcy Code and other similar laws in any other jurisdiction). The Parties agree that the other Party, as a licensee of such rights under this Agreement, shall retain and may fully exercise all of its rights and elections under any applicable insolvency statute, and that upon commencement of an Insolvency Event by or against the Rejecting Party, the other Party shall be entitled to a complete duplicate of or complete access to (as the other Party deems appropriate), any such intellectual property and all embodiments of such intellectual property. Such intellectual property and all embodiments thereof shall be promptly delivered to the other Party (a) upon any such commencement of a bankruptcy proceeding (or other Insolvency Event) upon written request therefor by the other Party, unless the Rejecting Party elects to continue to perform all of its obligations under this Agreement or (b) if not delivered under the foregoing clause (a), upon the rejection of this Agreement by or on behalf of the Rejecting Party, then upon written request therefor by the other Party. The provisions of this Section 3.6 shall be (i) without prejudice to any rights the other Party may have arising under any applicable insolvency statute or other applicable Laws and (ii) effective only to the extent permitted by applicable Laws.

3.7 Expansion of Territory. During the Term, the Parties may discuss amending the Territory based on additional business opportunities for Licensed Products outside of the then-current Territory. Any changes to the Territory shall require the written approval of both Parties.

3.8 Transfer of Know-How and Regulatory Materials.

- (a) Within thirty (30) days after the Effective Date, Skye shall provide Tautomer with complete and accurate written or electronic copies of (i) all Licensed Know-How and (ii) all Regulatory Materials regarding the Licensed Compound and Licensed Products that are necessary or useful for the Development or Commercialization of Licensed Products in the Territory, in each case (i) and (ii), that are in existence as of the Effective Date and Controlled by Skye or its Affiliates and on an ongoing basis thereafter, at least Calendar Quarterly, Skye shall promptly provide Tautomer with complete and accurate written or electronic copies of all Licensed Know-How and Regulatory Materials regarding the Licensed Compound and Licensed Products that are necessary or useful for the Development or Commercialization of Licensed Products in the Territory generated since the last such transfer under this Section 3.8(a), in each case, that are Controlled by Skye or its Affiliates. Notwithstanding the foregoing, Skye shall not be required to provide Tautomer with copies of any Licensed Know-How that solely relates to the Manufacture of Licensed Compound.
- **(b)** Within thirty (30) days after the Effective Date, Tautomer shall provide Skye with complete and accurate written or electronic copies of all Know-How included within the Tautomer IP in existence as of the Effective Date. On an ongoing basis thereafter, at least Calendar Quarterly, Tautomer shall promptly provide Skye with complete and accurate written or electronic copies of all Know-How included within the Tautomer IP generated since the last such transfer under this Section 3.8(b).

Article 4 GOVERNANCE

- **4.1 Joint Steering Committee.** Within thirty (30) days after the Effective Date, the Parties shall establish a joint steering committee (the "**Joint Steering Committee**" or "**JSC**") to provide oversight of strategic development, regulatory, commercial and safety matters and to coordinate, review and discuss the Parties' activities with respect to Development and Commercialization of Licensed Products. The responsibilities of the JSC shall include:
 - (a) facilitating communications and discussions between the Parties with respect to the Development Plan;
 - (b) reviewing, discussing and approving any proposed amendments to the Development Plan;
 - (c) providing a forum to discuss the Development of Licensed Products in the Field in the Territory;

- (d) providing a forum to discuss material Regulatory Materials for Licensed Products in the Territory and the progress of the Regulatory Approvals for Licensed Products in the Territory;
 - (e) providing a forum for discussion of safety governance matters relating to Licensed Compounds and Licensed Products;
 - (f) reviewing, discussing and approving the Commercialization Plan and reviewing and discussing any updates thereto
- (g) establishing additional committees as it deems necessary to manage the business under the Agreement, which committees shall have the responsibilities and authority as designated by the JSC (but not any authority that exceeds the authority of the JSC) and shall be subject to the direct oversight and control of the JSC;
 - (h) discussing proposed expansions of the Territory; and
- (i) performing such other functions as appropriate to further the purposes of this Agreement, as expressly set forth in this Agreement or as determined by the Parties in writing.
- **4.2 Limitation of Authority**. The JSC shall only have the powers expressly assigned to it in this Article 4 and shall not have the authority to: (a) modify or amend the terms and conditions of this Agreement or the Supply Agreement; (b) waive either Party's compliance with the terms and conditions of this Agreement or the Supply Agreement; (c) determine any issue in a manner that would conflict with the express terms and conditions of this Agreement or the Supply Agreement; (d) determine whether or not a Party has complied with any of its obligations under this Agreement or the Supply Agreement; or (e) make any decision or approve any matter that is expressly stated to require the mutual written agreement of the Parties or the written consent of one or both Parties.
- **4.3 Members.** Each Party shall appoint three (3) representatives to the JSC, each of whom will be an officer or employee of the applicable Party having sufficient seniority within such Party to make decisions arising within the scope of the JSC's responsibilities. The JSC may change its size from time to time by mutual agreement of the Parties; provided that the JSC always includes an equal number of representatives from each Party. Each Party may replace any of its representatives at any time upon written notice to the other Party. The JSC shall have a chairperson selected by Tautomer. The role of the chairperson shall be to convene and preside at the meetings of the JSC and to ensure the preparation of meeting minutes, but, except as set forth in Section 4.4, the chairperson shall have no additional powers or rights beyond those held by other JSC representatives.
- **4.4 Meetings.** The JSC shall meet at least one (1) time per Calendar Quarter, unless the Parties mutually agree in writing to a different frequency for such. Either Party may also call a special meeting of the JSC (by videoconference or teleconference) by at least ten (10) Business Days' prior written notice to the other Party in the event such Party reasonably believes that a significant matter must be addressed prior to the next regularly scheduled meeting, and such Party shall provide the JSC, no later than ten (10) Business Days prior to the special meeting,

with materials reasonably adequate to enable an informed decision. No later than ten (10) Business Days prior to any meeting of the JSC, the chairperson of the JSC shall prepare and circulate an agenda for such meeting; provided, however, that either Party may propose additional topics to be included on such agenda, either prior to or in the course of such meeting. The JSC may meet in person, by videoconference or by teleconference, as the Parties agree. Each Party shall bear the expense of its respective JSC members' participation in JSC meetings. Meetings of the JSC shall be effective only if at least one (1) representative of each Party is present or participating in such meeting. The chairperson of the JSC shall be responsible for preparing reasonably detailed written minutes of all JSC meetings that reflect, without limitation, all material decisions made at such meetings. The JSC chairperson shall send draft meeting minutes to each member of the JSC for review and approval within ten (10) Business Days after each JSC meeting. Such minutes shall be deemed approved unless one or more members of the JSC object to the accuracy of such minutes within ten (10) Business Days of receipt.

4.5 Decision Making. The JSC shall act by consensus. The representatives from each Party will have, collectively, one (1) vote on behalf of that Party. The JSC shall use reasonable efforts to seek consensus in its actions and decision-making process. If the JSC is unable to reach consensus on any issue for which it is responsible within twenty (20) days after first considering such issue, then either Party shall have the right to escalate the issue to the Senior Executives of each Party for discussion and resolution by good faith negotiations during a period of ten (10) Business Days. Any final decision mutually agreed to by the Senior Executives shall be conclusive and binding on the Parties. If such issue has not been resolved by the Senior Executives within ten (10) Business Days after being referred to them for resolution, then Tautomer shall have final decision-making authority with respect to matters primarily related to the Territory; provided that Tautomer shall not exercise its decision-making authority to take any action (a) that would reasonably be expected to have an adverse impact on the Licensed Compound, the Licensed Products or any other product containing or comprising the Licensed Compound outside of the Territory, (b) that is inconsistent with Skye's global development, commercialization or regulatory strategies with respect to the Licensed Compound, Licensed Products or any other product containing or comprising the Licensed Compound, (c) that would modify Skye's activities under the Development Plan or (d) that would breach or otherwise violate any agreement between Skye or any of its Affiliates and any Third Party.

Article 5 DEVELOPMENT

- 1.1 Overview. As between the Parties, Tautomer, itself and/or through its Affiliates and Sublicensees, will be responsible for all Development of the Licensed Products in the Field in the Territory, including all formulation, preclinical and clinical development and regulatory activities, and the associated costs of such activities. Tautomer shall conduct such tasks in a timely, professional manner and in compliance with the Development Plan and all applicable Laws, including GLP and GCP.
- 1.2 **Development Plan**. The initial written plan for the Development of Licensed Products in the Territory is set forth in <u>Schedule 5.2</u> (such plan, as amended from time to time in accordance with this Section 5.2, the "**Development Plan**"). Upon mutual agreement of the Parties, the Development Plan may include activities to be performed by Skye to support

17

Tautomer' Development of Licensed Compounds and Licensed Products in the Field in the Territory. Tautomer shall reimburse Skye for its costs and expenses in performing such activities in accordance with Section 8.2. From time to time, each Party may propose amendments to the Development Plan in consultation with the other Party and submit such proposed amended Development Plan to the JSC for review and approval. Once approved by the JSC, the amended Development Plan shall become effective.

- **5.1 Development Diligence**. Tautomer shall use Commercially Reasonable Efforts to Develop, including seeking Regulatory Approval for, Licensed Products in South Africa for both Chemotherapy Induced Nausea and Vomiting and Chronic Pain; provided that Tautomer shall not be required to Develop and seek Regulatory Approval for both Indications simultaneously.
- **5.2 Development Reports**. Tautomer will on a Calendar Quarterly basis provide Skye through the JSC with written reports summarizing its, its Affiliates' and its Sublicensees' Development of Licensed Compound and Licensed Products. Such reports shall contain sufficient detail to enable Skye to assess Tautomer' compliance with its Development obligations hereunder. Tautomer shall use reasonable efforts to generally include the following:
- (a) general information on Tautomer's and its Sublicensees' Development activities, including a summary of formulation development activities conducted during such Calendar Quarter and results of clinical studies and trials completed within such Calendar Quarter;
 - **(b)** a summary of planned Development activities;
 - (c) a timetable of planned and actual submissions for Regulatory Approvals;
- (d) information on all Regulatory Approvals obtained with respect to Licensed Products (notwithstanding any further reporting obligations in connection with the achievement of milestones); and
 - (e) a summary of meetings and significant communications with the Regulatory Authorities concerning Licensed Products.

The Parties shall discuss the status, progress and results of Development activities at JSC meetings. Tautomer shall also provide Skye with such additional information regarding the Development as Skye may reasonably request.

5.3 Records of Development Activities. Tautomer shall maintain, and shall cause its Affiliates and Sublicensees to maintain, records of its Development activities under this Agreement in sufficient detail and in good scientific manner appropriate for regulatory and patent purposes, which records shall be complete and accurate and shall fully and properly reflect all work done and results achieved in the performance of such activities under this Agreement. Tautomer's records maintained pursuant to this Section 5.5 shall not include or be commingled with records of activities conducted by Tautomer outside the scope of this Agreement. Tautomer shall maintain its records of activities conducted by Tautomer under this Agreement for at least three (3) years after the termination of this Agreement or such longer period required by

applicable Laws. Skye shall have the right no more than once per Calendar Year, during normal business hours and upon reasonable notice, to inspect any such records for the purpose of determining Tautomer's compliance with Section 5.3.

- 5.4 Subcontractors. Tautomer shall have the right to engage subcontractors for purposes of conducting its activities under this Agreement. Tautomer shall cause any subcontractor engaged by it to be bound by written obligations of confidentiality and non-use consistent with this Agreement prior to performing any activities. Tautomer shall cause its subcontractors to assign to Tautomer all IP discovered, made, generated, conceived or reduced to practice by such subcontractor in the course of performing such subcontracted work. Tautomer shall remain directly responsible for any of its obligations under this Agreement that have been delegated or subcontracted to any subcontractor and shall be directly responsible for the performance of its subcontractors.
- 5.5 Regulatory Responsibilities. As between the Parties, Tautomer will be responsible for all regulatory activities for Licensed Products in the Territory, including preparing and filing all Regulatory Materials for Licensed Products in the Territory and obtaining Regulatory Approvals to conduct clinical trials of Licensed Products in the Territory and to market Licensed Products in the Territory, at its sole cost and expense. Upon Tautomer's request, Skye will provide reasonable assistance to prepare Regulatory Materials related to Skye's Manufacture of Licensed Compound. Tautomer shall reimburse Skye for its costs and expenses in performing such activities in accordance with Section 8.2.
- **5.6** Review of Regulatory Submissions. Tautomer shall provide to Skye for review and comment English language drafts of all Regulatory Materials in the Territory for the Licensed Products no later than fifteen (15) days prior to the planned submission. Tautomer shall consider in good faith any comments received from Skye on such Regulatory Materials. In addition, Tautomer shall notify Skye of any material Regulatory Materials for the Licensed Products filed with or received from any Regulatory Authority and any other material documents, comments or other correspondences related thereto submitted to or received from any Regulatory Authority in the Territory and shall provide Tautomer with copies thereof as soon as reasonably practicable, but in all events within five (5) Business Days after submission or receipt thereof.
- 5.7 Right of Reference. Tautomer hereby grants to Skye the right of reference to any and all Regulatory Materials Controlled by Tautomer that relate to Licensed Products solely for the purpose of seeking, obtaining and maintaining Regulatory Approval of Licensed Products and products containing or comprising Licensed Compound outside of the Territory. Skye shall bear the reasonable costs and expenses of Tautomer associated with providing the right of reference pursuant to this Section 5.9. Tautomer will take such actions as may be reasonably requested by Skye to give effect to the intent of this Section 5.9 and to give Skye, its Affiliates, licensees and sublicensees the benefit of the rights of reference to Tautomer's Regulatory Materials as provided herein.
- 5.8 Regulatory Inspections. Tautomer shall promptly notify Skye in writing in the event that Tautomer becomes aware of any Regulatory Authority inspections relating to any Licensed Product in the Territory. Skye shall have the right to be present at any such inspections and shall have the opportunity to provide, review and comment on any responses that may be

required in connection with such inspections. In the event Tautomer does not receive prior notice of any such inspection, Tautomer shall notify Skye in writing as soon as practicable after such inspection and shall provide Skye with copies of all materials, correspondence, statements, forms and records received or generated pursuant to any such inspection. In addition to such obligations with respect to Regulatory Authority inspections, Tautomer shall promptly notify Skye in writing of any information it receives regarding any threatened or pending action or communication by or from any Third Party, including a Regulatory Authority, that may materially affect the Development, Manufacture, Commercialization or regulatory status of Licensed Products.

Adverse Events Reporting. No later than the first IND filing for a Licensed Product, Skye and Tautomer shall develop and enter into a written agreement for safety and pharmacovigilance procedures for the Parties with respect to Licensed Compounds and Licensed Products, such as safety data sharing and exchange and adverse events reporting and prescription events monitoring (the "Pharmacovigilance Agreement"). Such Pharmacovigilance Agreement shall (a) describe the obligations of both Parties with respect to the coordination of collection, investigation, reporting and exchange of information between the Parties concerning adverse events or any other safety issue of any significance and product quality and product complaints involving adverse events, in each case with respect to Licensed Compounds and Licensed Products and sufficient to permit each Party and its Affiliates, licensees or (sub)licensees to comply with its legal obligations with respect thereto; (b) be promptly updated if required by changes in applicable Laws; and (c) provide that (i) Tautomer shall maintain an adverse event database for clinical trials conducted in the Territory; (ii) Tautomer shall be responsible for (A) reporting to the applicable Regulatory Authorities in the Territory, all quality complaints, adverse events and safety data related to Licensed Products for all clinical trials conducted in the Territory, (B) responding, to safety issues and to all requests of Regulatory Authorities related to such safety issues with respect to Licensed Products in the Field in the Territory; (iii) Tautomer shall provide to Skye access to Tautomer's adverse event database for Licensed Products in the Territory; (iv) Tautomer shall promptly consult with Skye regarding all adverse event reports originating in the Territory and reasonably consider any input from Skye prior to finalizing adverse event reports for such events and/or making any submission to a Regulatory Authority regarding such events; (v) Skye shall maintain a global adverse event database for Licensed Compound; and (vi) Skye will provide Tautomer with access to Skye's global adverse event database regarding Licensed Compound and promptly provide Tautomer with any adverse event information regarding Licensed Compound in accordance with the Pharmacovigilance Agreement.

Article 6 MANUFACTURE

6.1 Manufacture of Licensed Compound. Skye, by itself or through the use of Contract Manufacturers, shall be solely responsible for the Manufacture and supply of, and Tautomer shall purchase from Skye, all of Tautomer's requirements for Licensed Compounds in the Territory pursuant to the Supply Agreement. The supply price for Licensed Compounds under the Supply Agreement shall be Skye's COGS plus twenty percent (20%).

- **6.2 Manufacture of Licensed Products**. Tautomer shall be solely responsible for Manufacturing (including packaging and labeling) Licensed Products in the Territory. Licensed Products for use in the Field in the Territory. Tautomer will Manufacture Licensed Products only in South Africa and shall only Manufacture Licensed Products using Licensed Compound supplied by or on behalf of Skye.
- 6.3 Supply Agreement. Promptly after the Effective Date (and in any event within sixty (60) days after the Effective Date), the Parties will negotiate in good faith and execute an agreement for the supply of Licensed Compound by Skye to Tautomer (the "Supply Agreement"). The Supply Agreement shall include commercially reasonable terms agreed to by the Parties. The Supply Agreement shall be consistent with, and shall be designed to permit Skye to comply with its obligations under, Skye's agreement(s) with its Contract Manufacturers for the Manufacture and supply of Licensed Compound and shall not impose on Skye additional obligations with respect to a Skye Contract Manufacturer's obligations to Skye with respect to such Licensed Compounds, and Skye shall be entitled to all disclaimers of warranties and limitations of liability with respect to the activities of such Skye Contract Manufacturer to which such Skye Contract Manufacturer is entitled with regard to supply of such Licensed Compounds.

Article 7 COMMERCIALIZATION

- 1.1 Commercialization Responsibilities. As between the Parties, Tautomer, itself and/or through its Affiliates and Sublicensees, will be solely responsible for all Commercialization activities for Licensed Product in the Territory, including, but not limited to, the marketing, strategy, pricing, promotion, physician targeting, reimbursement, distribution and sale of the Licensed Product in the Territory. Tautomer shall bear all of its costs and expenses incurred in connection with such Commercialization activities.
- **1.2** Commercial Diligence. Tautomer shall use Commercially Reasonable Efforts to Commercialize Licensed Products in South Africa for both Chemotherapy Induced Nausea and Vomiting and Chronic Pain.
- 1.3 Commercialization Plan and Reports. Tautomer shall submit a written strategic and tactical plan for the Commercialization of Licensed Products in the Field in the Territory (the "Commercialization Plan") to the JSC for review and approval. The Commercialization Plan with respect to a Licensed Product shall contain in reasonable detail the major Commercialization activities planned for such Licensed Product in the Territory. Tautomer shall deliver an initial draft of the Commercialization Plan to the JSC for its review at least six (6) months prior to the anticipated First Commercial Sale of a Licensed Product in the Territory. Tautomer shall provide the JSC with an updated Commercialization Plan on an annual basis.

Article 8 COMPENSATION

- Equity Purchase. Within sixty (60) days after the Effective Date (or such longer period as is necessitated by compliance with the restrictions contained herein), Tautomer shall purchase a minimum of Five Hundred Thousand Dollars (\$500,000) of shares of common stock Skye in the open market in compliance with all applicable laws and regulations. Tautomer shall appoint a broker reasonably agreeable to Skye as its agent to purchase common stock on behalf of Tautomer in the open market. It is the intention of the Parties that such purchases benefit from the safe harbor provided by Rule 10 b-18 ("Rule 10 b-18") promulgated by the Securities and Exchange Commission "SEC" under the Securities Act of 1934, as amended (the "Exchange Act"), and each Party acknowledges that Tautomer may be an "affiliated purchaser" of Skye, as such term is defined in Rule 10 b-18. Accordingly, Tautomer hereby agrees that, during the term of this Agreement, it shall not take, nor permit any person or entity under its control to take, any action that could jeopardize the availability of Rule 10 b-18 for such purchases of common stock pursuant to this Section 7.1. Tautomer shall enter into an agreement with such broker, in the form provided to and reasonably approved by Skye, evidencing such broker's obligation to make purchases of common stock hereunder in accordance with the requirements of this paragraph, including, without limitation, the timing, price and volume restrictions contained in Rule 10 b-18, taking into account the rules and practices of the principal exchange or trading market upon which the common stock is traded. Skye shall promptly notify Tautomer of the existence of any circumstances that render it advisable to suspend purchases under this Section 7.1 for any given period of time (including, without limitation, purchasers by Skye or any other "affiliated purchasers" (as defined in Rule 10 b-18) of Skye, distributions by Skye within the meaning of Regulation M under the Exchange Act or the possession by Tautomer of material non-public information), and upon receipt of Skye's direction to suspend purchases, Tautomer shall do so and notify its broker to such effect. Any suspension notice to the broker acting for Tautomer shall not include any other information about the nature of the circumstances giving rise to such suspension or its applicability to Tautomer or otherwise communicate any material nonpublic information about Skye or its securities to such broker. Tautomer shall promptly notify Skye in writing after making any purchases hereunder in accordance with monitoring procedures to be established by the Parties, including providing a daily email report confirming purchases of common stock to Skye indicating this specific price and number of shares purchased at each specific price for Tautomer for purposes of any required reporting by Skye in its filings with the SEC. Tautomer represents and warrants that it will publicly disclose the acquisition of any securities hereunder, as required, in accordance with Section 13 and Section 16 of the Exchange Act and that it will cooperate with Skye in all reasonable respects (including the execution of required documentation) to assist with and facilitate any public reporting obligations Skye may have under the Exchange Act or otherwise.
- 1.1 Reimbursement of Tautomer Development Expenses. Tautomer will reimburse Skye for all of Skye's internal and external costs and expenses plus a mark-up of [***] percent ([***]%) for the Development activities performed by Skye under the Development Plan and the regulatory support activities performed by Skye pursuant to Section 5.7 (collectively, "Skye Development Expenses"). Skye will invoice Tautomer for any Skye Development Expenses and Tautomer will pay such invoices within thirty (30) days of receipt.
- **1.2 Development Milestone Payments**. Within thirty (30) days after the first achievement by Tautomer, any of its Affiliates or Sublicensees of the following development milestone events, Tautomer shall notify Skye in writing and make the corresponding

development milestone payment to Tautomer. Each such development milestone payment shall be payable one time only.

Development Milestone Event	Payment
Complete Formulation of Licensed Product	\$[***]
Approval of Marketing Authorization Application for the first Indication	\$[***]
First Commercial Sale of Licensed Product for the first Indication	\$[***]

For purposes of this Section 8.3, "Complete Formulation of Licensed Product" shall be deemed to occur upon the first of (a) achievement of all requirements for formulation completion as set forth in the Development Plan or (b) initiation of the production of the first GMP-batch of Licensed Product.

8.2 Sales Milestone Payments. On an aggregate Net Sales basis for all Licensed Products in the Territory, Tautomer will pay to Skye the following one-time milestone payments within sixty (60) days after the first and only achievement of the applicable sales milestone event. For clarity, the milestone payments in this Section 8.4 shall be additive such that if multiple milestones below are met in the same Calendar Year, Tautomer shall pay all applicable payments to Skye for that Calendar Year.

Sales Milestone Event	Payment
Upon first achievement of annual Licensed Product Net Sales of \$[***] in the Territory	\$[***]
Upon first achievement of annual Licensed Product Net Sales of \$[***] in the Territory	\$[***]
Upon first achievement of annual Licensed Product Net Sales of \$[***] in the Territory	\$[***]
Upon first achievement of annual Licensed Product Net Sales of \$[***] in the Territory	\$[***]
Upon first achievement of annual Licensed Product Net Sales of \$[***] in the Territory	\$[***]

8.3 Royalties.

(a) Royalty Rates. Subject to Section 8.5(b), Tautomer will pay Skye tiered royalties on the aggregate, annual Net Sales of all Licensed Products in the Territory during the applicable Royalty Term, at the rates set forth below.

Portion of Annual Net Sales of Licensed Products in the Territory	Royalty Rate
Up to and including \$[***]	[***]%
Greater than \$[***] and less than or equal to \$[***]	[***]%
Greater than \$[***]	[***]%

(b) Royalty Term. Tautomer shall pay royalties under this Section 8.5, on a country-by-country and Licensed Product-by-Licensed Product basis, on Net Sales during the period of time beginning on the First Commercial Sale of such Licensed Product in such country and continuing until the later of: (i) the expiration of the last-to-expire of the Licensed Patents or patents within the Tautomer IP in the applicable country Covering such Licensed Product (including Licensed Compound contained in the Licensed Product), (ii) expiration of all

regulatory, marketing and data exclusivity applicable to such Licensed Product in such country or (iii) fifteen years after First Commercial Sale of such Licensed Product in such country (the "Royalty Term" in such country for such Licensed Product).

(c) Third Party Payments. If Tautomer in-licenses any Third Party Patents or trade secrets that are necessary to Develop, make, have made, use or sell Licensed Product in the Field in the Territory and the total royalties payable by Tautomer to the applicable Third Party for sales of Licensed Product exceed [***] per cent ([***]%) of Net Sales of such Licensed Product in such country, then the royalties payable to Skye in Section 8.5(a) shall be reduced on a Licensed Product-by-Licensed Product and country-by-country basis according to the following formula:

$$A \times 50\% \times (B \div C) = D$$

where:

- A = the percentage royalty payable by Tautomer to such Third Party for sales of such Licensed Product in such country that exceeds [***]% of Net Sales;
 - B = the royalty payable by Tautomer to Skye as set forth above for sales of such Licensed Product in such country;
 - C = the total royalty payable by Tautomer for sales of such Licensed Product in such country; and
- D = the amount by which the royalty payable to Skye as set forth above will be reduced for sales of such Licensed Product in such country;

provided always that the royalty payable to Skye for Net Sales of such Licensed Product in such country shall not be reduced by more than [***] percent ([***]%) below the royalty that would otherwise be owed pursuant to Section 8.5(a).

- (d) Compulsory Licenses. If a compulsory license is granted to a Third Party with respect to a Licensed Product in any country in the Field in the Territory with a royalty rate lower than the royalty rates provided by Section 8.5(a) (as adjusted per Section 8.5(c)), then the royalty rate to be paid by Tautomer on Net Sales in such country under Section 8.5(a) will be reduced to the rate payable by the compulsory licensee. For purposes of the foregoing, a "compulsory license" means, with respect to Licensed Product in a country or territory, a license, or rights granted to a Third Party by a governmental agency within such country or territory to sell or offer for sale Licensed Product in such country or territory under any Patents or Know-How Controlled by either Party or its Affiliates, without direct or indirect authorization from such Party or its Affiliates.
- (e) Royalty Reports and Payments. Within fifteen (15) days after the end of each Calendar Quarter during the Royalty Term, Tautomer shall deliver to Skye a statement, on a country-by-country and Licensed Product-by-Licensed Product basis, of the amount of gross sales and Net Sales of Licensed Products during the applicable Calendar Quarter, a calculation of the amount of royalty payment due on such sales for such Calendar Quarter, any applicable

royalty reductions under Section 8.5(c), and a revised calculation of the payment due after the application of such offsets. Concurrent with the delivery of the applicable quarterly report, Tautomer shall pay Skye in Dollars all royalties owed with respect to Net Sales for such Calendar Ouarter.

- 8.4 Non-Royalty Sublicensing Income. Subject to the terms of this Section 8.6, Tautomer shall pay Skye [***] percent ([***]%) of all Non-Royalty Sublicensing Income received by Tautomer or its Affiliates from Sublicensees. For clarity, to avoid double-counting of payments, Tautomer and its Affiliates shall have the right to credit the amounts of any milestone payment owed to Skye under Section 8.3 or Section 8.4 against any Non-Royalty Sublicensing Income arising from any payment received by Tautomer or its Affiliates from a Sublicensee for achievement of the substantially equivalent milestone event, prior to the calculation of the share of Non-Royalty Sublicensing Income payable to Skye. Within thirty (30) days after receipt of any Non-Royalty Sublicensing Income by Tautomer or its Affiliates, Tautomer shall notify Skye in writing and portion of Non-Royalty Sublicensing Income payable to Skye under this Section 8.6.
- 8.5 Currency; Exchange Rate. All payments to be made by Tautomer to Skye under this Agreement shall be made in Dollars by bank wire transfer in immediately available funds to a bank account designated by written notice from Skye. Tautomer shall be responsible for all wire transfer fees incurred in connection with its payments to Skye under this Agreement. The rate of exchange to be used in computing the amount of currency equivalent in Dollars shall be made at the average of the closing exchange rates reported in *The Wall Street Journal* (U.S., Eastern Edition) for the first, middle and last Business Days of the applicable reporting period for the payment due.
- **8.6 Late Payments.** If Skye does not receive payment of any sum due to it on or before the due date therefor, simple interest shall thereafter accrue on the sum due to Skye from the due date until the date of payment at a per-annum rate of prime (as reported in *The Wall Street Journal* (U.S., Eastern Edition)) plus two percentage points or the maximum rate allowable by applicable Law, whichever is less.
- **8.7 Records; Audits.** Tautomer shall maintain, and shall cause its Affiliates and Sublicensees to maintain, complete and accurate records in reasonably sufficient detail to permit Skye to confirm the accuracy of all payments owed to Skye under this Agreement and in compliance with Accounting Standards, consistently applied. Upon reasonable prior notice, such records shall be available during regular business hours for a period of ten (10) years from the end of the Calendar Year to which they pertain for examination, not more often than once each Calendar Year, to Skye or an independent certified public accountant selected by Skye and reasonably acceptable to Tautomer, for the sole purpose of verifying the accuracy of the financial reports furnished by Tautomer pursuant to this Agreement. Any such auditor shall enter into a confidentiality agreement with the Tautomer and shall not disclose Tautomer's Confidential Information, except to the extent such disclosure is necessary to verify the accuracy of the financial reports furnished by Tautomer or the amount of payments due by Tautomer under this Agreement. Any amounts shown to be owed but unpaid shall be paid, and any amounts showed to be overpaid will be refunded, within thirty (30) days from the completion of Skye's or the accountant's report. Skye shall bear the full cost of such audit unless such audit discloses and

underpayment by Tautomer of more than five percent (5%) of the amount due, in which case Skye shall bear the full cost of such audit.

8.8 Taxes.

- (a) All amounts to be paid by Tautomer hereunder are exclusive of Indirect Taxes. Applicable Indirect Taxes are to be duly calculated, accounted for and added to the fee amounts on the invoices to the extent required by applicable Laws. If any Indirect Taxes are owed, such Indirect Taxes shall be added to the amount owed and shall be paid by Tautomer. The Parties shall cooperate and exercise their commercially reasonable efforts to allow, to the extent possible under applicable Laws, recovery of any such VAT paid. Tautomer shall indemnify, defend, and hold Skye and its Affiliates harmless for any Indirect Taxes with respect to any amounts to be paid by Tautomer hereunder.
- withholdings for taxes as required by applicable Law in any country having jurisdiction. If applicable Law requires any such deduction or withholding, Tautomer shall timely remit the applicable deduction or withholding to the appropriate Governmental Authority, pay to Skye the applicable net amount after such deduction or withholding and secure and send to Skye the best available evidence of the payment of such withholding or similar tax within thirty (30) days following such payment. Notwithstanding the foregoing, if Tautomer (i) assigns or transfers this Agreement, (ii) exercises its rights or performs any of its obligations under this Agreement through an Affiliate, (iii) changes its domicile or performs any of its obligations under this Agreement establishment outside its country of incorporation or (iv) fails to comply with applicable Laws or filing or record retention requirements, and solely as a result of any such actions (i) (iv), the amount of any tax that Tautomer is required to deduct or withhold from a payment to Skye is increased, Tautomer shall be responsible for all such additional withholding taxes and shall increase the amount payable to Skye to the extent necessary to ensure that after making all required withholdings (including withholdings on additional amounts) Skye receives the same amount as it would have received had Tautomer not taken such action.
- (c) The Parties shall cooperate and exercise their commercially reasonable efforts to ensure that any withholding taxes imposed are reduced as far as possible under the provisions of applicable double tax treaties or other applicable Law. Prior to making any payment subject to deduction or withholding under Section 8.10(b), Tautomer shall provide a written notice to Skye of the amounts subject to deduction or withholding, and the legal basis therefor, and shall use commercially reasonable efforts to inform Skye of any forms, certificates or other items necessary to reduce or eliminate any withholding or similar taxes and provide Skye a reasonable opportunity to provide such forms, certificates or other items. The Parties shall furnish each other with the best available evidence on the application of double tax treaties applicable and of payment whenever Tautomer is to deduct such tax from any payments due.
- (d) The Parties shall use commercially reasonable efforts to provide, and to cause their respective Affiliates, subcontractors, Sublicensees, customers and applicable Third Parties to provide, any information and documentation reasonably requested by the other Party to obtain the benefits of (i) Section 250 of the Internal Revenue Code of 1986, as amended and the

applicable Treasury Regulations thereunder and/or (ii) any U.S. tax legislation enacted during the term of this Agreement that could provide a material tax benefit to either Party.

Article 9 INTELLECTUAL PROPERTY MATTERS

- 9.1 Ownership of Inventions. Inventorship of all inventions that are discovered, made, generated, conceived or reduced to practice in the course of activities performed under or contemplated by this Agreement shall be determined in accordance with the rules of inventorship under U.S. patent laws. As between the Parties, (a) Skye shall solely own all Compound IP, (b) each Party shall solely own all other Know-How, and intellectual property rights therein (including Patents Covering such Know-How), that is discovered, made, conceived or first reduced to practice solely by its or its Affiliates' employees, (sub)licensees, subcontractors or agents, excluding Compound IP and (c) the Parties shall jointly own all Know-How, and intellectual property rights therein (including Patents Covering such Know-How), that is discovered, made, conceived or first reduced to practice in connection with the Development, Manufacture or Commercialization of a Licensed Compound or Licensed Product under the Agreement where the joint inventors are (i) Skye's or its Affiliates employees, (sub)licensees (subject to the terms and conditions of existing agreements with such sublicensees), subcontractors or agents; and (b) Tautomer's or its Affiliate's employees, Sublicensees, subcontractors or agents, excluding Compound IP ("Joint IP"). All Patents claiming Joint IP shall be referred to herein as "Joint Patents." Except to the extent either Party is restricted by the licenses granted to the other Party under this Agreement, each Party shall be entitled to practice, license, assign and otherwise exploit the Joint IP and Joint Patents without the duty of accounting or seeking consent from the other Party.
- **9.2 Disclosure of Inventions**. Tautomer shall promptly disclose to Skye all inventions that are discovered, made, generated, conceived or reduced to practice by or on behalf of Tautomer, its Affiliates or Sublicensees under this Agreement, including all invention disclosures or other similar documents submitted to Tautomer by its or its Affiliates' employees, agents, or independent contractors relating thereto, and shall promptly respond to reasonable requests from Skye for additional information relating to such inventions.

9.3 Prosecution of Patents.

(a) By Skye. As between the Parties, Skye shall have the first right, but not the obligation, to prosecute and maintain, using counsel of its choice at its cost and expense, (i) the Licensed Patents and Joint Patents in the Territory and (ii) the Joint Patents and Patents included within the Tautomer IP (the "Tautomer Patents") outside of the Territory. Skye shall keep Tautomer reasonably informed of progress with regard to the prosecution and maintenance of any such Patents. In addition, Skye shall provide Tautomer with drafts of all proposed substantive filings and correspondence to any patent authority reasonably prior to the submission to such patent authority to the extent related to any such Patents for Tautomer's review and comment prior to the submission of such proposed filings and correspondence. Skye shall consider in good faith Tautomer's comments related to such Patents prior to submitting such filings and correspondence, provided that Tautomer provides such comments to Skye within thirty (30) days (or a shorter period reasonably designated by Skye if thirty (30) days is not

practicable given the filing deadline) of receiving the draft filings and correspondence from Skye. If Skye declines to prosecute and maintain any of (x) the Licensed Patents or Joint Patents in the Territory or (y) the Joint Patents or Tautomer Patents outside of the Territory, then it will promptly notify Tautomer in writing of its decision, and in any event, at least thirty (30) days in advance of any statutory bar or other deadline that would result in the loss of rights associated with such Licensed Patents, Joint Patents or Tautomer Patents. Tautomer may, upon written notice to Skye, assume such prosecution and maintenance in Skye's name or Tautomer' name, as applicable, and at Tautomer' sole expense.

- maintain, using counsel of its choice at its cost and expense, Tautomer Patents in the Territory. Tautomer shall keep Skye reasonably informed of progress with regard to the prosecution and maintenance of any Tautomer Patents. In addition, Tautomer shall provide Skye with drafts of all proposed substantive filings and correspondence to any patent authority reasonably prior to the submission to such patent authority to the extent related to any Tautomer Patents for Skye's review and comment prior to the submission of such proposed filings and correspondence. Tautomer shall consider in good faith Skye's comments related to such Patents prior to submitting such filings and correspondence, provided that Skye provides such comments to Tautomer within thirty (30) days (or a shorter period reasonably designated by Tautomer if thirty (30) days is not practicable given the filing deadline) of receiving the draft filings and correspondence from Tautomer. If Tautomer declines to prosecute and maintain any Tautomer Patent in the Territory, then it will promptly advise Skye of its decision, and in any event, at least thirty (30) days in advance of any statutory bar or other deadline that would result in the loss of rights associated with such Tautomer Patent. Skye may, upon written notice to Tautomer, assume such prosecution, defense and maintenance in Tautomer' name and at Skye's sole expense.
- (c) Cooperation. Each Party shall provide the other Party all reasonable assistance and cooperation, at the other Party's request and expense (and, for the avoidance of doubt, the requesting Party shall reimburse the assisting Party for all reasonable expenses in such assistance and cooperation), in the patent prosecution efforts provided above in this Section 9.3, including providing any necessary powers of attorney and executing any other required documents or instruments for such prosecution.

9.4 Enforcement of Licensed Patents.

- (a) Notification. If either Party becomes aware of (i) any existing or threatened infringement of the Licensed Patents (excluding any Licensed Patent that Covers only Licensed Compound), Joint Patents or Tautomer Patents in the Field in the Territory, which infringing activity involves the using, making, importing, exporting, offering for sale or selling Licensed Products or (ii) a declaratory judgment action in connection with any infringement described in clause (i) seeking an order that any claim in such Licensed Patent, Joint Patent or Tautomer Patent is invalid or unenforceable (each of (i) and (ii), a "Product Infringement"), it shall promptly notify the other Party in writing to that effect, and the Parties will consult with each other regarding any actions to be taken with respect to such Product Infringement.
- **(b) Enforcement Rights.** For any Product Infringement, each Party shall share with the other Party all information available to it regarding such actual or alleged

infringement. As between the Parties, Tautomer shall have the first right, but not the obligation, to bring an appropriate suit or take other action against any person or entity engaged in, or to defend against, such Product Infringement in the Territory, at Tautomer's cost and expense. If Tautomer does not, within sixty (60) days after its receipt or delivery of notice under Section 9.4(a), commence a suit to enforce the applicable Licensed Patents, Joint Patents or Tautomer Patents or take other action to terminate such Product Infringement, then Skye shall have the right, but not the obligation, to commence such a suit or take such an action or defend against such Product Infringement in the Territory at its own cost and expense; provided that prior to the expiration of such sixty (60)-day period Skye shall have the right, but not the obligation, to institute proceedings for interim injunctive relief against such Product Infringement. In such event, Tautomer shall take appropriate actions in order to enable Skye to commence a suit or take the actions set forth in the preceding sentence. Neither Party shall settle any such suit or action in any manner that would negatively impact the Licensed Patents or that would limit or restrict the ability of Tautomer to exploit Licensed Products anywhere in the Territory or limit or restrict the ability of Skye to exploit Licensed Compound and Licensed Products outside of the Territory without the prior written consent of the other Party.

- (c) Collaboration. Each Party shall provide to the enforcing Party reasonable assistance in such enforcement, at such enforcing Party's request and expense (and, for the avoidance of doubt, the enforcing Party shall reimburse the other Party for all reasonable expenses in such assistance), including joining such action as a party plaintiff if required by applicable Laws to pursue such action. The enforcing Party shall keep the other Party regularly informed of the status and progress of such enforcement efforts and shall reasonably consider the other Party's comments on any such efforts. The non-enforcing Party shall be entitled to separate representation in such matter by counsel of its own choice and at its own expense, but such Party shall at all times cooperate fully with the enforcing Party.
- (d) Expenses and Recoveries. The Party bringing or defending a claim, suit or action under Section 9.4(a) shall be solely responsible for any expenses incurred by such Party as a result of such claim, suit or action. If such Party recovers monetary damages in such claim, suit or action, such recovery shall be allocated first to the reimbursement of any expenses incurred by the Parties in such litigation, and any remaining amounts shall be allocated as follows: (i) if Tautomer is the enforcing or defending Party, the remaining amounts will be retained by Tautomer, except that any such amounts attributable to lost sales of Licensed Products shall be included in Net Sales subject to payments to Skye in accordance with Section 8.4 and Section 8.5 and all other amounts shall be treated as Non-Royalty Sublicensing Income subject to payment by Tautomer to Skye pursuant to Section 8.6, and (ii) if Skye is the enforcing or defending Party, Skye will retain eighty percent (80%) of any remaining amounts and remit twenty percent (20%) to Tautomer.
- 9.5 Patent Term Extensions. The Party responsible for prosecuting and maintaining Licensed Patents, Joint Patents or Tautomer Patents will be responsible for applying or having applied for patent term extensions, including supplementary protection certificates and any other extensions such as pediatric extensions, that are now available or become available during the Term and that become available directly as a result of the Regulatory Approval of a Licensed Product; provided that such Party will consult with the other Party with respect to such decisions and will consider the comments and concerns of the other Party in good faith.

9.6 Personnel Obligations. Prior to beginning work under this Agreement relating to any Development of a Licensed Product, each employee, agent or independent contractor of Tautomer or its Affiliates shall be bound by invention assignment obligations that are consistent with the obligations of Tautomer under this Agreement, including: (a) promptly reporting any invention, discovery, process or other intellectual property right; (b) assigning to Tautomer, as applicable, all of his or her right, title and interest in and to any invention, discovery, process or other intellectual property right; (c) cooperating in the preparation, filing, prosecution, maintenance and enforcement of any Patent; (d) performing all acts and signing, executing, acknowledging and delivering any and all documents required for effecting the obligations and purposes of this Agreement; and (e) complying with obligations of confidentiality and non-use consistent with those contained in this Agreement.

9.7 Trademarks.

- Trademarks. When selecting the Licensed Trademarks, Skye shall give good faith consideration to Tautomer's reasonable input. Tautomer shall not, and shall ensure that its Affiliates and Sublicensees will not, make any use of or attempt to register, the trademarks or house marks of Skye (including Skye's corporate name) or any trademark confusingly similar thereto. Skye shall have the first right, but not the obligation, to prosecute and maintain, using counsel of its choice at its cost and expense, the Licensed Trademarks in the Territory. Skye shall keep Tautomer reasonably informed of progress with regard to the prosecution and maintenance of any such trademarks. In addition, Skye shall promptly provide Tautomer with drafts of all proposed substantive filings and correspondence to any trademark authority to the extent related to any such trademarks for Tautomer's review and comment prior to the submission of such proposed filings and correspondence. Skye shall Tautomer provides such comments to Skye within thirty (30) days (or a shorter period reasonably designated by Skye if thirty (30) days is not practicable given the filing deadline) of receiving the draft filings and correspondence from Skye. If Skye declines to prosecute and maintain any Licensed Trademarks in any country in the Territory, then it will promptly advise Tautomer of its decision, however, at least thirty (30) days in advance of any statutory bar or other deadline that would result in the loss of rights associated with such Licensed Trademarks. Tautomer may, upon written notice to Skye, assume such prosecution and maintenance in Skye's name and at Tautomer's sole expense.
- (b) Each Party shall promptly notify the other Party in writing upon becoming aware of any infringement of the Licensed Trademarks in the Territory. As between the Parties, Tautomer shall have the first right, but not the obligation, to initiate a lawsuit or take other reasonable action to enforce the Licensed Trademarks against such infringement in the Territory. If Tautomer does not initiate a lawsuit or take other reasonable action intended to cause such infringement to cease and obtain remedies for the harm resulting therefrom within one hundred twenty (120) days of notice provided pursuant to this Section 9.7(b), then Skye shall have the right, but not the obligation, to initiate such lawsuit or take such other action, after providing thirty (30) days' notice to Tautomer. Any proceeds from such lawsuit after reimbursement of each Party's expenses shall be received by Tautomer; provided that, such received proceeds shall be deemed to be Net Sales and subject to payments to Skye in accordance with Section 8.4 and Section 8.5.

9.8 Recordal of Licenses. To the extent permitted and provided for under applicable Laws governing the jurisdiction in question, each Party shall have the right to have any license pursuant to this Agreement granted under a registered intellectual property right (including but not limited to any Licensed Patent and Licensed Trademark) in the Territory recorded on the register in question; and, to the extent that one Party (for purposes of this Section 9.8, the "Requesting Party") reasonably requires the action of the other Party (for purposes of this Section 9.8, the "Other Party") to effect such recordal, the Other Party shall perform any actions reasonably required by the Requesting Party to effect such recordal. To the extent that the Other Party incurs any costs in performing any action under this Section 9.8, unless otherwise agreed between the Parties, the Requesting Party shall bear the Other Party's reasonable costs of performing such action. Upon termination of any license that is recorded pursuant to this Section 9.8, whether as a result of termination of this Agreement or for any other reason, the Party that is the proprietor of the registered intellectual property right in question shall be entitled to have the recordal of the license deleted from the register.

Article 10 REPRESENTATIONS AND WARRANTIES; COVENANTS

- 10.1 Mutual Representations and Warranties. Each Party hereby represents and warrants to the other Party that, as of the Execution Date:
- (a) It is a company or corporation duly organized, validly existing, and in good standing under the Laws of the jurisdiction in which it is incorporated; and
- **(b)** (i) it has the corporate power and authority and the legal right to enter into this Agreement and perform its obligations hereunder; (ii) it has taken all necessary corporate action on its part required to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder; and (iii) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, valid and binding obligation of such Party that is enforceable against it in accordance with its terms.
- 10.2 Additional Representations and Warranties of Skye. Skye represents and warrants to Tautomer that, as of the Execution Date:
 - (a) Skye has the right to grant the license granted to Tautomer under Section 3.1;
- **(b)** Skye has not entered into any agreement with any Third Party that is in conflict with the rights granted to Tautomer under this Agreement;
- (c) Skye has not received any notice or threat from any Third Party asserting or alleging, and there is no basis for any assertion or allegation, that any research, manufacture or development of the Compound by Skye prior to the Execution Date misappropriated the intellectual property rights of such Third Party;
- (d) To Skye's knowledge, no Third Party is infringing or has infringed any Licensed Patents or has misappropriated any Licensed Know-How; and

- (e) Neither Tautomer nor any of its Affiliates have been debarred by a Regulatory Authority or any other Governmental Authority and is not the subject of a conviction by any such authority.
- 10.3 Additional Representations and Warranties of Tautomer. Tautomer represents and warrants to Skye that, as of the Execution Date:
 - (a) Tautomer has the right to grant the license and option to Skye it grants or purports to grant under Section 3.3; and
- **(b)** Neither Tautomer nor any of its Affiliates have been debarred by a Regulatory Authority or any other Governmental Authority and is not the subject of a conviction by any such authority.
 - **10.4 Skye Covenants**. Skye hereby covenants to Tautomer that:
 - (a) Skye will comply with applicable Laws in performing its activities under this Agreement;
- **(b)** During the Term, Skye will not enter into any contractual obligation with any Third Party that would conflict with the license granted to Tautomer under Section 3.1; and
- (c) In performing its activities under this Agreement, Skye shall not employ or engage any person who has been debarred by any Regulatory Authority or is the subject of debarment proceedings by a Regulatory Authority.
 - **10.5 Tautomer Covenants.** Tautomer hereby covenants to Skye that:
- (a) Tautomer will comply with applicable Laws in performing its activities under this Agreement, including with respect to the Development, Manufacture and Commercialization of Licensed Products.
- **(b)** During the Term, Tautomer will not enter into any contractual obligation with any Third Party that would conflict with the license or option granted to Skye under Section 3.3.
- (c) In performing its activities under this Agreement, Tautomer shall not employ or engage any person who has been debarred by any Regulatory Authority or is the subject of debarment proceedings by a Regulatory Authority, including, without limitation, any person who has been debarred or is the subject of debarment proceedings (i) under U.S. law, including under 21 U.S.C. § 335a, or any foreign equivalent thereof or (ii) that are related to the development or approval of any therapeutic product, any regulation of any therapeutic product. Tautomer shall notify Skye promptly upon becoming aware that any of its or its Affiliates' employees or consultants has been debarred or is the subject of debarment proceedings by any Regulatory Authority, or engages in any conduct or activity which could lead to debarment or a debarment proceeding.

(d) During the Term, Tautomer and its Affiliates shall not, by themselves or with or through any Third Party, Manufacture Licensed Compound or Develop, Manufacture, Commercialize or otherwise exploit any product containing or comprising Licensed Compound except for Licensed Products.

10.6 Mutual Covenants. Each Party hereby covenants to the other Party that:

- (a) it shall not, in the performance of this Agreement, perform any actions that are prohibited by applicable anti-corruption laws (including the provisions of the United States Foreign Corrupt Practices Act of 1977, as amended, the U.S. Travel Act, the U.S. Domestic Bribery Statute and all other applicable laws and regulations in jurisdictions in which the Party engages in business that govern corruption, bribery, kickbacks and ethical business conduct) ("Anti-Corruption Laws") or Export Control Laws;
- (b) neither such Party nor any of its Affiliates will, in connection with the exercise of such Party's rights or performance of its obligations under this Agreement, directly or indirectly through Affiliates or Third Parties, pay, promise, offer to pay, authorize the payment of, accept or solicit, any money or give any promise, offer to give or authorize the giving of anything of value (including, but not limited, any corrupt payment, gratuity, emolument, bribe, kickback, improper gift, hospitality or benefit) to or from a public official or entity, or any other person for the purpose of obtaining or retaining business for or with, or directing business to, any person, including such Party and its Affiliates;
- (c) in connection with the exercise of such Party's rights or performance of its obligations under this Agreement, except as permitted by applicable government license or authorization, it shall not engage in any transactions or dealings with (including export, reexport or transfer any items to) (i) any country or territory that is subject to an embargo by the U.S. government (currently, Cuba, Iran, North Korea, Syria and the Crimea, Donetsk People's Republic and Luhansk People's Republic regions of Ukraine) or (ii) any entity or individual subject to sanctions, including being identified on any list of designated and prohibited parties maintained by the United States and other applicable jurisdictions (including, but not limited to, the List of Specially Designated Nationals and Blocked Persons, the Foreign Sanctions Evaders List and the Sectoral Sanctions Identifications List, which are maintained by the Office of Foreign Assets Control of the U.S. Treasury Department and the Entity List, Denied Persons List and Unverified List, which are maintained by the Bureau of Industry and Security of the U.S. Commerce Department);
- (d) it shall maintain records (financial and otherwise) and supporting documentation related to the subject matter of this Agreement in order to document or verify compliance with the provisions of this Section 10.6, and upon request of the other Party, up to one time per Calendar Year and upon reasonable advance written notice, shall provide the other Party or its representative with access to such records for purposes of verifying compliance with the provisions of this Section 10.6; and
- (e) such Party shall immediately notify the other Party, to the extent permitted by the applicable Law, if such Party has any information or suspicion that there may be a violation of Anti-Corruption Laws or Export Control Laws in connection with the exercise of such Party's rights or performance of such Party's obligations under this Agreement.

10.7 Disclaimer. EXCEPT AS EXPRESSLY STATED IN THIS ARTICLE 10, (A) NO REPRESENTATION, CONDITION OR WARRANTY WHATSOEVER IS MADE OR GIVEN BY OR ON BEHALF OF TAUTOMER OR SKYE; AND (B) ALL OTHER CONDITIONS AND WARRANTIES WHETHER ARISING BY OPERATION OF LAW OR OTHERWISE ARE HEREBY EXPRESSLY EXCLUDED, INCLUDING ANY CONDITIONS AND WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR NON-INFRINGEMENT.

Article 11 INDEMNIFICATION

- 1.1 Indemnification by Skye. Skye shall defend, indemnify, and hold Tautomer and its Affiliates and their respective officers, directors, employees, and agents (the "Tautomer Indemnitees") harmless from and against any and all damages or other amounts payable to a Third Party claimant, as well as any reasonable attorneys' fees and costs of litigation incurred by such Tautomer Indemnitees, resulting from any claims, suits, proceedings or causes of action brought by such Third Party (collectively, "Claims") against such Tautomer Indemnitee to the extent arising from or based on (a) the breach of any of Skye's obligations, agreements, covenants, representations or warranties under this Agreement, (b) the Development, Manufacture (excluding Manufacture of Licensed Compound for Tautomer, which will be addressed in the Supply Agreement), use, handling, storage, sale or other disposition of any Licensed Compound or Licensed Product by Skye or its Affiliates (except to the extent attributable to or arising out of the actions of Tautomer, its Affiliates or Sublicensees or its or their employees or contractors) or (c) the gross negligence, willful misconduct or violation of applicable Laws by Skye or any of its Affiliates under this Agreement. The foregoing indemnity obligation shall not apply to the extent that any Claim arises from or is based on any activity for which Tautomer is obligated to indemnify the Skye Indemnitees under Section 11.2.
- 1.2 Indemnification by Tautomer. Tautomer shall defend, indemnify, and hold Skye and its Affiliates and their respective officers, directors, employees and agents (the "Skye Indemnitees") harmless from and against damages or other amounts payable to a Third Party claimant, as well as any reasonable attorneys' fees and costs of litigation incurred by such Skye Indemnitees, resulting from any Claims against such Skye Indemnitee to the extent arising from or based on (a) the Development, Manufacture or Commercialization of Licensed Products by or on behalf of Tautomer or its Affiliates or Sublicensees (except (i) to the extent attributable to or arising out of the actions of Skye, its Affiliates or (sub)licensees or its or their employees or contractors or (ii) to the extent attributable to the Manufacture of Licensed Compound for Tautomer, which will be addressed in the Supply Agreement), (b) the breach of any of Tautomer's obligations, agreements, covenants, representations or warranties under this Agreement or (c) the gross negligence, willful misconduct or violation of applicable Laws by Skye or any of its Affiliates under this Agreement. The foregoing indemnity obligation shall not apply to the extent that any Claim arises from or is based on any activity for which Skye is obligated to indemnify the Tautomer Indemnitees under Section 11.1.
- 1.3 Indemnification Procedures. If either Party is seeking indemnification under Sections 11.1 or 11.1 (the "Indemnified Party"), it shall promptly notify the other Party (the "Indemnifying Party") in writing of the Claim giving rise to the obligation to indemnify

pursuant to such Section as soon as reasonably practicable after receiving notice of the Claim (but in no event later than thirty (30) days after receipt such notice; provided that any failure to make or delay in making such notification shall not relieve the Indemnifying Party of its obligations hereunder except to the extent the Indemnifying Party is materially prejudiced by such failure or delay). The Indemnifying Party shall have the right to assume the defense of any such Claim for which it is obligated to indemnify the Indemnified Party. The Indemnified Party shall cooperate with the Indemnifying Party as the Indemnifying Party may reasonably request, and at the Indemnifying Party's cost and expense. The Indemnified Party shall have the right to participate, at its own expense and with counsel of its choice, in the defense of any Claim that has been assumed by the Indemnifying Party. Neither Party shall have the obligation to indemnify the other Party in connection with any settlement made without the Indemnifying Party's written consent, which consent shall not be unreasonably withheld or delayed. The Indemnifying Party may not enter into any compromise or settlement unless (a) such compromise or settlement imposes only a monetary obligation on the Indemnifying Party and includes as an unconditional term thereof, the giving by each claimant or plaintiff to the Indemnified Party of a release from all liability in respect of such claim; or (b) the Indemnified Party consents to such compromise or settlement, which consent will not be unreasonably withheld, conditioned or delayed unless such compromise or settlement involves (i) any admission of legal wrongdoing by the Indemnified Party, (ii) any payment by the Indemnified Party that is not indemnified under this Agreement or (iii) the imposition of any equitable relief against the Indemnified Party. If the Parties cannot agree as to the application of Section 11.1 or 11.1 as to any Claim, pending resolution of the dispute pursuant to Article 14, the Parties may conduct separate defenses of such Claims, with each Party retaining the right to claim indemnification from the other Party in accordance with Section 11.1 or 11.2 upon resolution of the underlying Claim.

- **1.4 Limitation of Liability.** NEITHER PARTY SHALL BE LIABLE TO THE OTHER FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, PUNITIVE, OR INDIRECT DAMAGES ARISING FROM OR RELATING TO ANY BREACH OF THIS AGREEMENT, REGARDLESS OF ANY NOTICE OF THE POSSIBILITY OF SUCH DAMAGES. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS SECTION 11.3 IS INTENDED TO OR SHALL LIMIT OR RESTRICT THE INDEMNIFICATION RIGHTS OR OBLIGATIONS OF ANY PARTY UNDER SECTION 11.1 OR 11.1, OR DAMAGES AVAILABLE FOR A PARTY'S BREACH OF CONFIDENTIALITY OBLIGATIONS IN ARTICLE 12.
- 1.5 Insurance. Tautomer shall procure and maintain insurance, including product liability insurance, with respect to its activities hereunder that is consistent with normal business practices of prudent companies similarly situated at all times during which any Licensed Product is being clinically tested in human subjects or commercially distributed or sold. Tautomer shall provide Skye with evidence of such insurance upon request and shall provide Skye with written notice at least sixty (60) days prior to the cancellation, non-renewal or material changes in such insurance. Such insurance shall not be construed to create a limit of Tautomer' liability with respect to its indemnification obligations under this Article 11.

Article 12 CONFIDENTIALITY

- 12.1 Confidentiality. During the Term and for a period of seven (7) years thereafter, all Confidential Information disclosed by or on behalf of a Disclosing Party to a Receiving Party hereunder shall be maintained in confidence by the Receiving Party and shall not be disclosed to any Third Party or used for any purpose other than as expressly provided for in this Agreement (which includes the exercise of any rights or the performance of any obligations hereunder) or otherwise used or disclosed in any manner that would constitute a violation of any applicable Laws, including any applicable Export Control Laws or securities Laws. The foregoing confidentiality and non-use obligations shall not apply to any portion of the Disclosing Party's Confidential Information that the Receiving Party can demonstrate by competent written proof:
- (a) was already known to the Receiving Party or its Affiliate, other than under an obligation of confidentiality, at the time of disclosure by the Disclosing Party;
- (b) was generally available to the public or otherwise part of the public domain at the time of its disclosure to the Receiving Party;
- (c) became generally available to the public or otherwise part of the public domain after its disclosure and other than through any act or omission of the Receiving Party in breach of this Agreement;
- (d) was disclosed to the Receiving Party or its Affiliate by a Third Party who has a legal right to make such disclosure and who did not obtain such information directly or indirectly from the Disclosing Party; or
- (e) was independently discovered or developed by the Receiving Party or its Affiliate without access to or aid, application or use of the Disclosing Party's Confidential Information.

Any combination of features or disclosures shall not be deemed to fall within the foregoing exclusions merely because individual features are published or available to the general public or in the rightful possession of the Receiving Party unless the combination itself and principle of operation are published or available to the general public or in the rightful possession of the Receiving Party.

- 12.2 Authorized Disclosure. Notwithstanding the obligations set forth in Section 12.1, the Receiving Party may disclose the Disclosing Party's Confidential to the extent:
- (a) such disclosure is reasonably necessary to its employees, agents, consultants, contractors or (sub)licensees on a need-to-know basis for the sole purpose of performing its obligations or exercising its rights under this Agreement; provided that in each case, the disclosees are bound by written obligations of confidentiality and non-use consistent with those contained in this Agreement; provided that the Receiving Party shall be responsible for any breach of this Article 12 by its employees, agents, consultants, contractors or (sub)licensees; or

- (b) such disclosure is reasonably necessary for filing and prosecuting Patents in accordance with Section 9.3;
- (c) such disclosure is reasonably necessary for filing Regulatory Materials with a Regulatory Authority or obtaining, maintaining or expanding any Regulatory Approval;
- (d) such disclosure is reasonably necessary to comply with applicable Laws, including regulations promulgated by applicable security exchanges, court order, administrative subpoena or order; provided that the Party subject to such Laws shall promptly notify the other Party of such required disclosure and shall use reasonable efforts to obtain, or to assist the other Party in obtaining, a protective order preventing or limiting the required disclosure.

Notwithstanding the foregoing, if a Party is required to make a disclosure of the other Party's Confidential Information pursuant to Section 12.2(d), such Party shall notify the other Party of such required disclosure as far in advance as reasonably practicable (and in no event less than fifteen (15) Business Days prior to the anticipated date of disclosure provided that such time is reasonably available to such Party) to provide the non-disclosing Party opportunity to review and comment upon the disclosure.

12.3 Return of Confidential Information. Any reproduction by a Party or its representatives of any of the other Party's Confidential Information shall be and remain the Disclosing Party's property and shall contain all confidential or proprietary notices or legends that appear on the original. All Confidential Information of the Disclosing Party (including all copies of it) shall always remain the Disclosing Party's property. Upon the expiration or termination of this Agreement or upon the Disclosing Party's request, the Receiving Party and its representatives shall promptly destroy (and certify in writing the destruction of) all Confidential Information of the Disclosing Party (including all copies, records and other embodiments of it in any medium), together with any derivative information, including notes, analyses, summaries and other tangible materials representing the Disclosing Party's Confidential Information. Notwithstanding the foregoing, the Receiving Party may retain one (1) copy of the Confidential Information in the Receiving Party's secure archives for the sole purpose of monitoring compliance with its continuing obligations under this Agreement, and the Receiving Party shall not be obligated to delete any electronic back-up or archival storage copies made in accordance with such Receiving Party's normal practices solely for purposes of disaster recovery and compliance with its records retention practices. Notwithstanding the destruction of Confidential Information, the Receiving Party will continue to be bound by its nondisclosure and non-use obligations surviving under this Agreement.

12.4 Publicity; Terms of Agreement.

(a) The Parties agree that the terms of this Agreement are the Confidential Information of both Parties, subject to the authorized disclosure provisions set forth in this Section 12.3 or Section 12.2. In addition, a Party may disclose such terms to the extent reasonably necessary to be disclosed to any bona fide potential or actual investor, acquiror, merger partner or (sub)licensee for the sole purpose of evaluating an actual or potential investment, acquisition, merger or (sub)licensing agreement; provided that in connection with such disclosure, such Party shall inform each disclosee of the confidential nature of such Confidential Information and ensure that each such disclosee is contractually obligated to treat

such Confidential Information as confidential in accordance with commercially reasonable industry standards for such disclosures.

The Parties have mutually approved a joint press release attached hereto as Exhibit A with respect to this Agreement, and either Party may make subsequent public disclosures of the contents of such press release. Subject to the foregoing, neither Party shall issue any other public announcement, press release or other public disclosure regarding the terms of this Agreement without the other Party's prior written consent, such consent not to be unreasonably withheld, delayed or conditioned, except for any such announcement, release or disclosure that is, in the opinion of counsel of the Party proposing to issue such announcement, release or disclosure, required by applicable Laws or the rules of a stock exchange on which the securities of the Party proposing to issue such announcement, release or disclosure are listed (or to which an application for listing has been submitted). If a Party is, in the opinion of its counsel, required by applicable Laws or the rules of a stock exchange on which its securities are listed (or to which an application for listing has been submitted) to make such a public disclosure, such Party shall, to the extent legally feasible, submit the proposed disclosure in writing to the other Party as far in advance as reasonably practicable under the circumstances, and shall in good faith consider any comments received from the other Party. Under such circumstances, the releasing Party shall not be obligated to delay making any such press release or public communication beyond the time when the same is required to be made in order to facilitate comment by the other Party. Neither Party shall be required to seek the permission of the other Party to repeat any information regarding the terms of this Agreement or any amendment hereto that has already been publicly disclosed by such Party or by the other Party, in accordance with this Section 12.4(b); provided that such information remains accurate as of such time and provided the frequency and form of such disclosure are reasonable. Notwithstanding anything to the contrary in this Article 12, either Party may issue press releases or make public disclosures regarding (i) the commencement, progress, status, completion and key results of each clinical trial for any Licensed Product or (ii) regulatory updates and receipt of Regulatory Approvals for Licensed Product. In addition, either Party may disclose any payments paid to or received by either Party in respect of the achievement of any milestone events in the event, in the opinion of counsel of the Party issuing such disclosure, such disclosure is required by applicable Laws or the rules of a stock exchange on which the securities of the Party issuing such disclosure are listed (or to which an application for listing has been submitted). For clarity, the Party making any disclosure hereunder shall have the final say over the contents of such disclosure.

(c) The Parties acknowledge that either or both Parties may be obligated to file under applicable Laws a copy of this Agreement with the U.S. Securities and Exchange Commission or other Governmental Authorities. Each Party shall be entitled to make such a required filing; provided that it requests confidential treatment of the commercial terms and sensitive technical terms hereof and thereof to the extent such confidential treatment is reasonably available to such Party. In the event of any such filing, each Party will provide the other Party with a copy of this Agreement marked to show provisions for which such Party intends to seek confidential treatment and shall reasonably consider and incorporate the other Party's reasonable comments thereon to the extent consistent with the legal requirements, with respect to the filing Party, governing disclosure of material agreements and material information that must be publicly filed.

12.5 Scientific Publications. Either Party may publish or present data and/or results generated under this Agreement and relating to a Licensed Compound or Licensed Product in scientific journals and/or at scientific conferences, subject to this Section 12.5. Each Party shall provide the other Party with the opportunity to review any proposed abstract, manuscript or presentation which discloses data and/or results generated under this Agreement and relating to a Licensed Compound or Licensed Products (each, a "Publication") by delivering a copy thereof to the other Party no less than forty-five (45) days before the intended submission for publication or presentation (the "Review Period"). Each Party agrees that it will not submit or present any Publication (a) until the other Party has provided written comments during such Review Period on the material in such Publication or (b) until the applicable Review Period has elapsed without written comments from the other Party. If a publishing Party receives written comments from the other Party during the applicable Review Period, it shall consider the comments of the other Party in good faith, but will retain the sole authority to submit the manuscript for Publication; provided that the other Party agrees (i) to delete any Confidential Information of the other Party that the other Party identifies for deletion in its written comments and (ii) to delay such Publication for a period of up to an additional ninety (90) days after the end of the applicable Review Period to enable the other Party to draft and file Patents with respect to any subject matter to be made public in such Publication and to which the other Party has the right to file such Patents. Each Party agrees to acknowledge the contributions of the other Party and its employees in all Publications as scientifically appropriate. Each Party will provide to the other a copy of the final version of each Publication.

Article 13 TERM AND TERMINATION

- 13.1 Term. This Agreement shall become effective on the Effective Date and, unless earlier terminated pursuant to this Article 13, shall remain in effect on Licensed Product-by-Licensed Product and a country-by-country basis, until the expiration of the Royalty Term of such Licensed Product in such country (the "Term"); provided that Article 1, Article 2, Article 11, Article 14 and Article 15 and Section 10.1, Section 10.2, 10.3, Section 10.6, Section 10.7, Section 12.1, Section 12.2, Section 12.3, Section 12.4, Section 13.1 and Section 13.6 shall be effective upon the Execution Date. Upon the expiration of the Royalty Term for a Licensed Product in a particular country, the licenses granted by Skye to Tautomer under Section 3.1(a) with respect to such Licensed Product and such country shall become fully-paid, royalty free and non-exclusive.
- **13.2** Unilateral Termination by Tautomer. Tautomer may terminate this Agreement in its entirety for any or no reason upon ninety (90) days' written notice to Skye.
- 13.3 Unilateral Termination by Skye. Skye may terminate this Agreement in its entirety upon ninety (90) days' written notice to Tautomer if (a) (i) Skye undergoes a Change of Control (such written notice to be provided within ninety (90) days after the occurrence of such Change of Control) or (ii) Skye intends to grant to a Third Party a worldwide, exclusive license for the Licensed Compound outside the Field and (b) Skye pays to Tautomer a buy-out fee ("Buy-Out Fee") determined in accordance with Schedule 13.3.

- 13.4 Termination for Material Breach. Notwithstanding any other remedies and sanctions available to it, either Party may terminate this Agreement in its entirety upon written notice to the other Party in the event the other Party materially breaches this Agreement and fails to cure such breach within ninety (90) days (or thirty (30) days with respect to a breach of payment obligations) after receipt of written notice of breach from the non-breaching Party.
- 13.5 Termination by Either Party for Bankruptcy. Either Party may terminate this Agreement in its entirety upon written notice to the other Party in the event that (a) a case is commenced by or against such other Party under applicable bankruptcy, insolvency or similar laws, which case, if commenced against (not by) such other Party, is not dismissed within ninety (90) days of the commencement thereof, (b) such other Party files for bankruptcy, reorganization, liquidation, receivership or similar proceedings, (c) such other Party assigns all or a substantial portion of its assets for the benefit of creditors, (d) a receiver or custodian is appointed for such other Party's business, (e) a substantial portion of such other Party's business is subject to attachment or similar process or (f) anything analogous to any of the events described in the foregoing clauses (a) through (e) occurs under the laws of any applicable jurisdiction (each of (a) to (f), an "Insolvency Event").
- 13.6 Termination for Failure to Fulfill Suspensive Condition. If the Suspensive Condition has not been fulfilled prior to expiration of the Exchange Control Approval Period, then at any time thereafter either Party shall have the right to terminate this Agreement in its entirety upon written notice to the other Party referencing this Section 13.6. Upon termination of this Agreement pursuant to this Section 13.6, this Agreement shall, except as otherwise provided in this Section 13.6, be of no further force and effect, the Term shall never commence and neither Party shall have any further rights or liability hereunder. For clarity, upon termination of this Agreement pursuant to this Section 13.6 neither Party shall have any liability to the other Party for any provision of this Agreement that did not come into effect as of the Execution Date. The following sections shall survive such termination of this Agreement: Article 1, Article 14 and Article 15 and Section 10.7, Section, 11.1, Section 11.2, Section 11.3, Section 11.4, Section 11.5 (last sentence only), Section 12.1, Section 12.2, Section 12.3, Section 12.4 and Section 13.6.

13.7 Effect of Termination.

- (a) Termination of License to Tautomer. Upon the termination (but not expiration) of this Agreement for any reason, all rights and licenses granted to Tautomer under this Agreement shall terminate and revert to Skye.
- **(b) Reversion of Rights.** Upon termination of this Agreement by Tautomer pursuant to Section 13.2 or by Skye pursuant to Section 13.3, 13.4 or 13.5:
- (i) License Grant to Skye. Tautomer hereby grants to Skye, effective upon such termination, an exclusive, irrevocable, perpetual, royalty-free, fully paid-up, sublicensable (through multiple tiers) license under the Tautomer IP and Tautomer's rights in the Joint IP to Develop, Manufacture and Commercialize Licensed Compound and Licensed Products in the Field.

- (ii) Tautomer Covenant. Tautomer and its Affiliates shall not, either themselves or with or through any Third Party, further Develop, Manufacture or Commercialize any Licensed Product in the Territory.
- (iii) Regulatory Materials; Data. Tautomer shall promptly transfer and assign to Skye, at no cost to Skye, all Regulatory Materials and Regulatory Approvals for the Licensed Products, all data from non-clinical and clinical studies conducted by or on behalf of Tautomer, its Affiliates or Sublicensees on the Licensed Products, and all pharmacovigilance data (including all adverse event databases) on the Licensed Products
- (iv) Inventory. At Skye's request, Tautomer shall deliver to Skye all inventory (if any, and to the extent applicable) of Licensed Products owned by Tautomer (or its Affiliate) and in Tautomer' (or its Affiliates) possession or control, and Skye shall reimburse Tautomer for its cost of goods for manufacturing or having manufactured such inventory.

(v) Transition Assistance.

- (A) Tautomer shall, upon Skye's request, supply the Licensed Product in its then-current form to Skye at cost (without markup) for a reasonable period of time until Skye establishes an alternative supplier, and in any event for at least twelve (12) months, and reasonably assist Skye in establishing an alternative supplier for such Licensed Products.
- **(B)** Upon Skye's request, Tautomer shall assign or sublicense to Skye any license agreements with respect to the Licensed Products in the Territory and any agreements or arrangement with Third Party vendors pertaining to the Development, Manufacture or Commercialization of Licensed Products in the Territory.
- (C) Tautomer shall, at Skye's request, provide reasonable technical assistance, including assistance with any inquiries and correspondence with Regulatory Authorities relating to any Licensed Product, for a period of twelve (12) months after the effective date of termination, and transfer all Know-How within the Tautomer IP relating to the Licensed Products.
- (D) If at the time of the notice of termination, Tautomer is conducting any clinical trials for a Licensed Product, then, at Skye's election on a trial-by-trial basis: (A) Tautomer shall fully cooperate with Skye to transfer the conduct of all such clinical trials to Skye, according to a transition plan to be developed by the Parties, and Skye shall assume any and all liability for such clinical trials after the effective date of such transition (except to the extent arising from any act or omission by Tautomer, its Affiliates or their respective employees, agents and contractors), provided that Tautomer shall continue to bear all costs and expenses incurred in connection with the conduct of such clinical trials until the completion of such transition; or (B) Tautomer shall, at its expense, orderly wind down the conduct of any such clinical trial that is not assumed by Skye under clause (A).
- **(E)** In addition to the foregoing, Tautomer shall use reasonable efforts with respect to those activities for which it is responsible to ensure orderly transition and uninterrupted Development, Manufacturing and Commercialization of Licensed Products by

Skye and to enable Skye to enter into an agreement with a Third Party to continue these activities with minimal disruption and delay.

13.8 Survival. Termination or expiration of this Agreement shall not affect any rights or obligations of the Parties under this Agreement that have accrued prior to the date of termination or expiration. Notwithstanding anything to the contrary, the following provisions shall survive any expiration or termination of this Agreement: [Article 1, Article 12, Article 14 and Article 15 and Sections 5.5, 8.7, 8.8, 8.9, 8.10, 9.1, 10.7, 11.1, 11.2, 11.3, 11.4, 13.6 and 13.8] [Note to draft: To be finalized prior to execution.]

Article 14 DISPUTE RESOLUTION

14.1 Disputes. In the event of any dispute, controversy or claim arising out of, in connection with or related to this Agreement, or the interpretation, performance, enforcement, validity, breach or termination thereof, then upon the written request of either Party, the matter shall be referred to the Senior Executives, who shall meet in a good faith effort to resolve the dispute. Any final decision mutually agreed to by the Senior Executives shall be conclusive and binding on the Parties. If the Senior Executives are not able to agree on the resolution of any such dispute within thirty (30) days (or such other period of time as mutually agreed by the Senior Executives) after such dispute was first referred to them, then such dispute shall be resolved, subject to Sections 4.5 and 8.9, pursuant to the provisions of Section 14.2.

14.2 Arbitration.

- (a) All disputes that are not resolved pursuant to Section 14.1 shall be finally settled by arbitration in accordance with the rules of the International Chamber of Commerce (the "Rules") in effect at the time of the arbitration, except as may be modified herein, by one (1) or more arbitrators selected in accordance with the Rules, provided that in the event there are three (3) arbitrators, each Party shall nominate one (1) arbitrator and the Party-nominated arbitrators shall nominate the third arbitrator within twenty (20) days after appointment of the second arbitrator. The seat, or legal place, of arbitration shall be New York, New York, USA. The language to be used in the arbitral proceedings shall be English. Any dispute regarding the scope, effect or applicability of the agreement to arbitrate or the propriety of commencing arbitration, shall be determined by the arbitrator(s) under this Section 14.2.
- **(b)** The arbitrators shall render its final award, no later than twelve (12) months following transmission of the arbitration file to the arbitrator(s), unless the Parties jointly request an extension, or the arbitrator(s) determine in a reasoned decision that the interest of justice or the complexity of the case requires that such limit be extended.
- (c) The arbitration shall be kept confidential and any nonpublic information provided in the arbitration, including any submissions, witness statements, expert reports or documents submitted in the arbitration or orders or decisions of the arbitrator(s) shall not be disclosed to any non-party except the arbitrator(s), the International Chamber of Commerce, the Parties, their counsel, experts, witnesses, accountants and auditors, insurers and reinsurers and any other person necessary to the conduct of the arbitration. The arbitrator(s) shall issue appropriate protective orders to safeguard each Party's Confidential Information. Except as

required by applicable Law, no Party shall make (or instruct the arbitrator(s) to make) any public announcement with respect to the proceedings or an order, decision or award of the arbitrator(s) without prior written consent of the other Party. The existence of any dispute submitted to arbitration, and the award, shall be kept in confidence by the Parties and the arbitrators, except (i) as required in connection with the enforcement of such award, (ii) as otherwise required by applicable Law or regulation requiring a Party to fulfil a legal duty or to protect or pursue a legal right, (iii) for actions to challenge the award, (iv) with the consent of both Parties or (v) where such information is already in the public domain other than as a result of a breach of this clause. No award or procedural order made in the arbitration shall be published.

- (d) Except as set forth herein or as determined by the arbitrators, each Party shall bear its own legal fees for the arbitration. The costs, fees and expenses of the arbitrator(s) and the International Chamber of Commerce will be borne equally by the Parties.
- (e) The arbitration award shall be final and binding on the Parties and the Parties undertake to carry out any award without delay. Any award shall be promptly paid in Dollars free of any tax, deduction or offset, and any costs, fees or taxes incident to enforcing the award shall, to the maximum extent permitted by applicable Law, be charged against the Party resisting enforcement. Judgment upon the award may be entered in any court of competent jurisdiction and the Parties hereby consent to the jurisdiction of such court for purposes of enforcement of such award.
- 14.3 Injunctive Relief. Nothing contained in this Agreement shall deny either Party the right to seek interim equitable relief from a court of competent jurisdiction in the context of a bona fide emergency or prospective irreparable harm, and such an action may be filed and maintained notwithstanding any ongoing discussions between the Parties or any ongoing arbitration proceeding.

Article 15 MISCELLANEOUS

- 1.1 Entire Agreement; Amendment. This Agreement, including the Exhibits hereto, sets forth the complete, final and exclusive agreement and all the covenants, promises, agreements, warranties, representations, conditions and understandings between the Parties hereto with respect to the subject matter hereof and supersedes, as of the Execution Date, all prior and contemporaneous agreements and understandings between the Parties with respect to the subject matter hereof, including the Confidentiality Agreement. The foregoing shall not be interpreted as a waiver of any remedies available to either Party as a result of any breach, prior to the Execution Date, by the other Party of its obligations under the Confidentiality Agreement. There are no covenants, promises, agreements, warranties, representations, conditions or understandings, either oral or written, between the Parties other than as are set forth in this Agreement. No subsequent alteration, amendment, change or addition to this Agreement shall be binding upon the Parties unless reduced to writing and signed by an authorized officer of each Party.
- 1.2 Force Majeure. Both Parties shall be excused from the performance of their obligations under this Agreement to the extent that such performance is prevented by force

majeure and the nonperforming Party promptly provides notice thereof to the other Party. Such excuse shall be continued so long as the condition constituting force majeure continues and the nonperforming Party takes reasonable efforts to remove the condition. For purposes of this Agreement, force majeure shall include conditions beyond the control of the Parties, including an act of God, war, civil commotion, terrorist act, labor strike or lock-out, epidemic, failure or default of public utilities or common carriers, destruction of production facilities or materials by fire, earthquake, storm or like catastrophe, and failure of plant or machinery (provided that such failure could not have been prevented by the exercise of skill, diligence and prudence that would be reasonably and ordinarily expected from a skilled and experienced person engaged in the same type of undertaking under the same or similar circumstances). Notwithstanding the foregoing, a Party shall not be excused from making payments owed hereunder because of a force majeure affecting such Party. If a force majeure persists for more than ninety (90) days, then the Parties will discuss in good faith the modification of the Parties' obligations under this Agreement in order to mitigate the delays caused by such force majeure.

1.3 Notices. Any notice required or permitted to be given under this Agreement shall be in writing, shall specifically refer to this Agreement, and shall be addressed to the appropriate Party at the address specified below or such other address as may be specified by such Party in writing in accordance with this Section 15.3, and shall be deemed to have been given for all purposes (a) when received, if hand-delivered or sent by confirmed electronic mail or a reputable courier service, or (b) five (5) Business Days after mailing, if mailed by first class certified or registered airmail, postage prepaid, return receipt requested.

If to Skye:

Skye Bioscience, Inc. 11975 El Camino Real, Suite 305 San Diego, CA 92130 United States Attn: Punit Dhillon

With a copy to (which shall not constitute notice):

Cooley LLP 10265 Science Center Drive San Diego, CA 92121 United States Attn: Charity R. Williams

If to Tautomer:

Woodmead North Office, 54 Maxwell Drive, Block B, Woodmead, 2191, Gauteng Province, South Africa

Attn: Mr. Martin Magwaza

15.1 Assignment. Neither Party may assign or transfer this Agreement or any rights or obligations hereunder without the prior written consent of the other, except that a Party may make such an assignment or transfer without the other Party's consent to its Affiliates or to a Third Party successor to substantially all of the business of such Party to which this Agreement relates, whether in a merger, sale of stock, sale of assets or other transaction. Any successor or assignee of rights and/or obligations permitted hereunder shall, in writing to the other Party, expressly assume performance of such rights and/or obligations. Any permitted assignment shall be binding on the successors of the assigning Party. Any assignment or attempted assignment by either Party in violation of the terms of this Section 15.4 shall be null, void and of no legal effect.

15.2 Headings; Interpretation.

- (a) The captions to the several Articles, Sections, subsections, Exhibits and Schedules hereof are not a part of this Agreement, but are merely for convenience to assist in locating and reading the several Articles and Sections hereof. Unless the context requires otherwise or otherwise specifically provided, (a) all references herein to Articles, Sections, Schedules or Exhibits shall be construed to refer to Articles, Sections, Schedules and Exhibits of this Agreement and (b) reference in any Section to any subclauses are references to such subclauses of such Section.
- (b) The definitions of the terms herein shall apply equally to the singular and plural forms of the terms defined and where a word or phrase is defined herein, each of its other grammatical forms shall have a corresponding meaning. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine, and neuter forms. The word "will" shall be construed to have the same meaning and effect as the word "shall". The word "any" shall mean "any and all" unless otherwise clearly indicated by context. The words "including", "includes", "include", "for example", and "e.g." and words of similar import will be deemed to be followed by the words "without limitation." The word "or" is disjunctive but not necessarily exclusive. The words "hereof", "herein" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement.
- 15.3 Independent Contractors. It is expressly agreed that Skye and Tautomer shall be independent contractors and that the relationship between the two Parties shall not constitute a partnership, joint venture or agency. Neither Skye nor Tautomer shall have the authority to make any statements, representations or commitments of any kind, or to take any action, which shall be binding on the other Party, without the prior written consent of the other Party. The Parties (and any successor, assignee, transferee, or Affiliate of a Party) shall not treat or report the relationship between the Parties arising under this Agreement as a partnership for applicable tax purposes without the prior written consent of the other Party unless required by applicable Laws.
- **15.4 Waiver.** The waiver by either Party hereto of any right hereunder, or of any failure of the other Party to perform, or of any breach by the other Party, shall not be deemed a waiver of any other right hereunder or of any other breach by or failure of such other Party whether of a similar nature or otherwise.

- 15.5 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of New York without giving effect to any choice of law principles that would require application of the laws of a different jurisdiction.
- 15.6 Severability. If any one or more of the provisions contained in this Agreement is held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby, unless the absence of the invalidated provision(s) adversely affects the substantive rights of the Parties. The Parties shall in such an instance use their best efforts to replace the invalid, illegal or unenforceable provision(s) with valid, legal and enforceable provision(s) which, insofar as practical, implement the purposes of this Agreement.
- **15.7 Cumulative Remedies.** No remedy referred to in this Agreement is intended to be exclusive, but each shall be cumulative and in addition to any other remedy referred to in this Agreement or otherwise available under law.
- 15.8 Waiver of Rule of Construction. Each Party has had the opportunity to consult with counsel in connection with the review, drafting and negotiation of this Agreement. Accordingly, the rule of construction that any ambiguity in this Agreement shall be construed against the drafting Party shall not apply.
- 15.9 Translations. This Agreement is in the English language only, which language shall be controlling in all respects, and all versions hereof in any other language shall be for accommodation only and shall not be binding upon the Parties. All communications and notices to be made or given pursuant to this Agreement, and any dispute proceeding related to or arising hereunder, shall be in the English language. If there is a discrepancy between any translation of this Agreement and this Agreement, this Agreement shall prevail.
- **15.10 Further Actions.** Each Party agrees to execute, acknowledge and deliver such further instruments, and to do all such other acts, as necessary or appropriate in order to carry out the purposes and intent of this Agreement.
- **15.11** Counterparts. This Agreement may be executed in two or more counterparts by original signature, facsimile or PDF files, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument

[Signature page follows]

In Witness Whereof, each Party has caused this Agreement to be executed on its behalf by its duly authorized representatives as of the Execution Date.

Skye Bioscience, Inc.

Tautomer Bioscience (Pty) Limited

By: /s/ Punit Dhillon

By: /s/ Martin S. Magwaza

Name: Punit Dhillon

Name: Martin S. Magwaza

Title: CEO & Chairman

Title: Founder & CEO

Exhibit A Joint Press Release

Schedule 1.58 Licensed Compound [***]

Schedule 5.2 Initial Development Plan

[***]

Schedule 13.3 Determination of Buy-Out Amount

[***]

OFFER TO SELL

UNIT 104 – 9295 198TH STREET LANGLEY, BC

THIS OFFER TO SELL made the 23 day of November, 2023. ("Offer" or "Agreement")

TO: COLLIERS MACAULAY NICOLLS INC.

200 Granville Street, 19th Floor Granville Square Vancouver, BC V6C 2R6

(hereinafter called the "Agent")

BETWEEN:

TAB LABS INC.

Unit D107, 19720 94A Avenue Langley, BC V1M 3B7

(hereinafter called the "Purchaser")

And:

AVALITE SCIENCES INC.

104-9295 198th Street Langley BC V1M 3J9

(hereinafter called the "Vendor", together with the Purchaser, the "Parties")

We the undersigned hereby offer to sell the property municipally described as Unit 104 – 9295 198th Street, Langley BC having a legal description and more particularly known as:

PID: 015-551-768 STRATA LOT 4, SECTION 34, TOWNSHIP 8, NEW WESTMINSTER DISTRICT STRATA PLAN NW3095 (hereinafter the "**Property**")

on the following terms and conditions:

1. PURCHASE PRICE

The PURCHASE PRICE shall be the sum of One Million Four Hundred Ninety Nine Thousand Dollars (\$1,499,000) (the 'Purchase Price') payable as follows:

- (a) REFUNDABLE DEPOSIT: As a portion of the Purchase Price and within two (2) business days of acceptance of this Offer, the sum of Twenty Five Thousand Dollars (\$25,000) (the "**Refundable Deposit**"), by way of a cheque or bank draft shall be paid to Colliers Macaulay Nicolls Inc., in trust, to be held as contemplated in this Offer;
- (b) FINAL DEPOSIT: The Purchaser shall, within two (2) business days of (i) the Purchaser's Conditions Precedent (as defined below) being satisfied or waived by the Purchaser; and (ii) the satisfaction of the Mutual Condition (as defined below), pay to the Vendor an additional sum of Fifty Thousand Dollars (\$50,000) (the "Final Deposit" and together with the Refundable Deposit, the "Deposit"). Subject to Clause 1(f), the Deposit will be non-refundable and paid by way of a

V	Ρ
---	---

cheque or bank draft payable to Colliers Macaulay Nicolls Inc., in trust, and to be held as contemplated by this Offer; and

(c) BALANCE OF PURCHASE PRICE: The balance of the Purchase Price, shall be payable by way of certified trust cheque or wire transfer to the Vendor or the Vendor's solicitors at the time of closing as herein provided.

The Deposit will be held in accordance with the following:

- (d) The Refundable Deposit will be returned to the Purchaser if the Purchaser does not satisfy or waive the Purchaser's Conditions Precedent set out in Clause 3 or the Purchaser and the Vendor do not satisfy the Mutual Condition set out in Clause 4;
- (e) The Deposit will be applied to the Purchase Price on the Closing Date if the Vendor and the Purchaser complete the sale and purchase of the Property;
- (f) The Deposit will be immediately refunded to the Purchaser if the Vendor is in default of its obligations to complete the sale of the Property hereunder, without prejudice to any other right or remedy of the Purchaser; and
- (g) The Deposit will be retained by and absolutely forfeit to the Vendor as liquidated damages if the Purchaser, through the Purchaser's default, fails to complete the purchase of the Property after having waived or satisfied the Purchaser's Conditions Precedent set out in Clause 3 and the satisfaction of the Mutual Condition set out in Clause 4 as the Vendor's sole and exclusive remedy;

provided that \$10.00 of the Deposit shall, in any event, be non-refundable to the Purchaser in accordance with Clause 3.

2. VENDOR RESPONSIBILITIES

The Vendor will deliver to the Purchaser all information in its possession related to the Property (the "Project Documents"), which includes:

- (a) Strata documents, including but not limited to meeting minutes (Council meetings, annual general meetings and special general meetings), budgets, and bylaws for the last two (2) fiscal years
- (b) Building plans
- (c) Service contracts
- (d) proposed list of Equipment

3. PURCHASER'S CONDITION PRECEDENT

This Offer is subject to the following condition precedent being satisfied or waived by the Purchaser forty-five (45) days from mutual acceptance of this Agreement (the "Subject Waiver Date"):

- (a) Review and approval of the Project Documents as noted in Clause 2;
- (b) The Purchaser conducting and approving any inspections of the Property required to complete this transaction; and



(c) The Purchaser receiving approval from its Board of Directors to complete this transaction;

(the "Purchaser's Conditions Precedent").

If the Purchaser fails to notify the Vendor in writing that the Purchaser's Conditions Precedent have been satisfied or waived within the time specified, or by such time as may be subsequently agreed, then this Agreement will become null and void and the Refundable Deposit will be returned to the Purchaser in full, except \$10.00 which shall be retained by the Vendor as valuable consideration for these Purchaser's Conditions Precedent.

The Purchaser's Conditions Precedent are for the sole benefit of the Purchaser. The Purchaser has the right to waive the Purchaser's Conditions Precedent at its discretion within the time stipulated and proceed with the transaction herein contemplated.

In the event that the Purchaser does not purchase the Property in accordance with the terms hereof, the Purchaser will indemnify and save harmless the Vendor from and against any loss, cost or damage of any nature whatsoever sustained by the Vendor as a result of the Purchaser exercising its rights under Sub-clause 3(b), reasonable wear and tear excepted; provided, however, that Purchaser shall have no liability under this provision to the extent such loss, cause or damage arose from or is related to the gross negligence or willful misconduct of Vendor or anyone acting on behalf or under the authority of Vendor. Notwithstanding anything to the contrary herein including, the Purchaser's obligation to indemnify set forth in this Clause 3 shall survive termination of this Agreement for any reason whatsoever.

4. MUTUAL CONDITION

This Offer is subject to the following condition precedent being satisfied or waived by both the Purchaser and the Vendor by the Subject Waiver Date:

(a) The Vendor and the Purchaser agreeing upon the itemized list of equipment (the "Equipment") to be sold by the Vendor to the Purchaser, agreeing upon the purchase price for such Equipment and entering into and executing a separate purchase and sale agreement for the Equipment.

(the "Mutual Condition").

The Mutual Condition is for the mutual benefit of the Purchaser and the Vendor and may not be waived unilaterally by either party. If the Mutual Condition is not satisfied within the time specified, or by such time as may be subsequently agreed, then this Agreement will become null and void and the Refundable Deposit will be returned to the Purchaser in full.

5. CLOSING

The closing shall take place at the office of the Purchaser's solicitor on January 15th, 2024 (the 'Closing Date"). The Purchaser shall have possession of the Property, subject to the Permitted Liens, following payment of the Purchase Price to the Vendor or the Vendor's solicitor on the Closing Date. All adjustments both incoming and outgoing with respect to rent, taxes, utilities and other items normally adjusted between a vendor and a purchaser with respect to the Property, shall be made as of 12:01 a.m. on the Closing Date. If the current year taxes are not known on the Closing Date, taxes will be adjusted on the basis that current year taxes will be 10% higher than the previous years' taxes but the parties will readjust ten days after determination of the actual current year taxes. If any item subject to adjustment cannot be determined on or before the Closing Date, an



estimate shall be made jointly by the Vendor and the Purchaser, acting reasonably, for purposes of closing and a final adjustment shall be made when the particular item can be determined. All claims for readjustments must be made within a twelve month period following the Closing Date and after the expiry of such twelve month period, the adjustments made by the parties prior to the expiry of such period shall be final and binding.

The obligation of the Purchaser to complete the purchase and sale transactions herein contemplated will be subject to the condition that all of the obligations herein of the Vendor will have been materially performed by the Vendor in accordance with this Agreement except as may be waived in writing by the Purchaser, and that all representations, warranties, covenants and agreements in this Agreement or any document delivered in connection with the transactions herein contemplated will be materially true and correct at and as of Closing Date as though such representations, warranties, covenants and agreements were made at and as of the Closing Date.

6. VENDOR REPRESENTATION, WARRANTIES AND COVENANTS

The Vendor represents, warrants, and covenants where applicable, that:

- (a) the Vendor is the owner of the Property and has the power to dispose of the same free and clear of all encumbrances save for reservations in the original grant from the Crown and those encumbrances outlined in the title search attached hereto as Schedule A (the "Permitted Liens");
- (b) there will be no tenancies, license agreements or other occupancy or use agreements affecting the Property at the time of closing;
- (c) the Vendor has received no notice of any expropriation of the Property;
- (d) the Vendor has received no notice of any outstanding orders against the Property or notices threatening orders by the Fire Warden, Health Department, Building or Engineering Departments of the Township of Langley or any other department or government agency which has jurisdiction over the Property;
- (e) the Property will be free and clear of all liens, charges and encumbrances on the Closing Date save for the Permitted Liens;
- (f) subject to and without in any way limiting the other representations and warranties herein contained, the Property will be maintained and transferred to the Purchaser in the same condition as inspected by the Purchaser pursuant to Sub-Clause 3(b);
- (g) the Vendor is a resident of Canada within the meaning of the *Income Tax Act*;
- (h) at the Closing Date, the Vendor shall deliver a certificate executed by a senior officer (without personal liability) of the Vendor familiar with the Property stating that at such time the above representations and warranties continue to be materially true and correct or, if not true and correct, stating details thereof;
- (i) Vendor has not entered into any material commitments or agreements with any governmental authorities or agencies affecting the Property, and Vendor has not received any written notice of an intention to revoke any certificate of occupancy, license, or permit issued in connection with the Property:
- (j) the Vendor has not, in connection with the Vendor's use of the Property, been charged with, received notice of non-compliance, or convicted of any offence for

V	Ρ
---	---

non-compliance with any provision term of the *Environmental Management Act* SBC 2003 c.53 (the "**EMA**"), or been fined or otherwise sentenced or settled any prosecution short of conviction, and there are no notices of judgment or commencement of proceedings of any nature and to the Vendor's knowledge, without investigation, the Vendor has not ever been investigated relating to any breach or alleged breach of the EMA;

- (k) the Vendor has not been required by any governmental authority to perform any environmental closure, decommissioning, rehabilitation, restoration or post-remedial investigations, on, about, or in connection with the Property other than in accordance with or to comply with the EMA;
- (I) to the Vendor's knowledge, without investigation, the Property has not been contaminated by any Hazardous Waste, as defined in the EMA, in amounts in excess of those permitted under the EMA; and
- (m) there will be no service, maintenance, operating, and other contracts with respect to the Property at the time of closing.

The above representations and warranties shall survive the Closing Date and shall continue in full force for twelve (12) months. The Vendor agrees unconditionally to indemnify, protect and save harmless the Purchaser and their respective officers, directors, employees, agents, successors and assigns, (collectively, the "Purchaser Indemnified Parties") against and in respect of any loss, cost or damage of any nature whatsoever sustained, including legal fees and disbursements on a full indemnify basis, which may be made or brought against any of the Purchaser Indemnified Parties, or which any of the Purchaser Indemnified Parties may suffer, incur or be required to pay, directly or indirectly as a result of, in respect of or arising out of any misrepresentation, inaccuracy, incorrectness or breach of any representation or warranty made by the Vendor contained in this Agreement, or in any other agreement, schedule certificate or other document required to be entered into or delivered by the Vendors pursuant to this Agreement.

7. PURCHASER'S REPRESENTATIONS, WARRANTIES AND COVENANTS

The Purchaser represents, warrants, and covenants where applicable, that:

- (a) the Purchaser is a corporation duly incorporated and validly existing under the laws of the Province of British Columbia;
- (b) the Purchaser has the power, authority and capacity to enter into this Agreement, to perform its obligations hereunder and to carry out the transactions contemplated by this Agreement;
- (c) the execution and delivery of this Agreement and the completion of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of the Purchaser;
- (d) the Purchaser is registered for GST purposes;
- (e) the Purchaser is not a non-resident of Canada within the meaning of the Income Tax Act (Canada); and
- (f) at the time of closing, the Purchaser shall deliver a certificate executed by a senior officer (without personal liability) of the Purchaser stating that at such time the above representations and warranties continue to be true and correct or, if not true and correct, stating details thereof.



The above representations and warranties shall survive the Closing Date and shall continue in full force for twelve (12) months. The Purchaser shall give written notice to the Vendor of each breach of a representation or warranty, together with details thereof, with ten (10) business days after becoming aware of the breach and in any event by no later than the date that is twelve (12) months after the Closing Date.

8. PURCHASER'S ACKNOWLEDGEMENTS AND AGREEMENTS

The Purchaser hereby acknowledges and agrees that:

- (a) Notwithstanding anything contained herein to the contrary, the Purchaser is purchasing the Property in an "as is/where is" condition, subject only to the covenants, representations and warranties made by the Vendor in this Agreement. The Purchaser further acknowledges that, subject only to the covenants, representations and warranties made by the Vendor in this Agreement, it is the Purchaser's responsibility to satisfy itself with respect to the Property and all matters relating to or affecting the Property, including without limitation, the state of repair of the Property, the zoning of the Property and the environmental and physical condition of the Property;
- (b) there are no warranties or representations, expressed or implied, by statute or otherwise, given by the Vendor as to the condition (environmental or otherwise), suitability or fitness of the Property, except those specifically set out in this Agreement;
- (c) the provisions of this section shall not merge on, but shall survive, the Closing Date.

9. ASSIGNMENT

It is understood and agreed that the Purchaser shall have the right to assign its rights and obligations under this Agreement or direct a transfer of the Property to another party with the consent of the Vendor, such consent not to be unreasonably withheld, but in such case the Purchaser shall remain liable in respect of its obligations hereunder. If this Agreement is assigned, the Vendor will be entitled to any profit resulting from an assignment of the Agreement by the Purchaser or any subsequent assignee. Notwithstanding the foregoing, the Purchaser will have the right to assign its rights under this Agreement to any entity which remains, at all times up to and including the Closing Date, an Affiliate of the Purchaser (which has the meaning ascribed to it in the Business Corporations Act (British Columbia)), or any successor or related entity that may arise via amalgamation or merger or other similar transaction involving Purchaser or any of its Affiliates, without the consent of the Vendor, provided that:

- (a) the Purchaser will deliver written notice to the Vendor of any such assignment;
- (b) the Purchaser will remain fully liable to the Vendor for the performance by any such Affiliate of the obligations of the Purchaser under the Agreement and will not be released from the performance hereof; and
- (c) the Affiliate enters into an agreement with the Vendor assuming the rights and obligations of the Purchaser under this Agreement.



10. FINTRAC

FINTRAC is the Financial Transactions and Reporting Analysis Center of Canada. Bill C25 has passed and came into effect June 23, 2008 and relates to the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act* to which the real estate industry is bound by. The Vendor and Purchaser acknowledge and agree that the Agent is bound by the above legislation and will assist the Agent in being compliant with the relevant legislation. This includes providing the Agent with proof of identity of Individuals and/or Corporations pertaining to this transaction. The Agent will keep all such materials and information private and confidential at all times as required by regulatory and corporate privacy requirements.

11. GENERAL

- (a) Time shall be of the essence of this Agreement.
- (b) This Agreement shall be governed by and construed in accordance with the laws of the Province of British Columbia.
- (c) It is understood that there are no representations, warranties, guarantees, promises or agreements other than those contained in this Agreement.
- (d) The Deposit will be placed in a non interest-bearing trust account.
- (e) It is acknowledged and agreed by the Vendor and the Purchaser that the GST and any PST, sales or similar tax imposed pursuant to provincial legislation ("Sales Tax") shall be an amount payable by the Purchaser in addition to the Purchase Price. The Purchaser shall pay all applicable property or land transfer tax, GST and Sales Tax in respect of the transaction contemplated by this Agreement and shall pay any such taxes collectible by the Vendor at closing on the Closing Date. If a registrant under the Excise Tax Act, the Purchaser shall provide to the Vendor a certificate to that effect on the Closing Date including an indemnity for any penalties, claims or losses arising from the failure of the Purchaser to account for GST (the "GST Certificate") and shall pay or account to Canada Revenue Agency pursuant to the Excise Tax Act, all GST payable in respect of the sale and purchase of the Property pursuant to this Offer. If the Purchaser is not a registrant under the Excise Tax Act it shall remit to the Vendor all GST payable in respect of the sale and purchase of the Property pursuant to this Offer.
- (f) The Purchaser hereby indemnifies and holds harmless the Vendor from and against, and shall pay and reimburse the Vendor for any and all claims, liabilities, tax, penalties, interest, costs and legal and other expenses incurred, directly or indirectly in respect of any PST, GST, Sales Tax or land transfer tax payable by the Purchaser which the Vendor fails to collect from the Purchaser at closing on the Closing Date.
- (g) Each of the Agent, Purchaser and Vendor shall execute and deliver all such further documents and do such further acts and things as may be reasonably required from time to time to give effect to this Agreement.
- (h) This Offer may only be amended by an agreement in writing by the parties hereto.
- (i) The Purchaser and its respective principals, agents and affiliates shall not disclose any information regarding this transaction, including, without limitation, the existence of the Offer, or the price and terms of the sale contemplated

V	Ρ
---	---

hereby, or other property information, except to its respective accountants, solicitors, lenders and advisors.

- (j) This Offer and the agreements, instruments and other documents entered into pursuant to this Offer set forth the entire agreement and understanding of the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect to the matters herein and there are no oral or written agreements, promises, warranties, terms, conditions, representations or collateral agreements whatsoever, express or implied, other than those contained in this Offer.
- (k) This Offer will enure to the benefit of and be binding upon the heirs, executors, administrators, legal representatives, successors and permitted assigns of the parties, as applicable.

12. LIABILITY OF DAMAGE TO PROPERTY

The Property shall be at the risk of the Vendor until the time of the closing of the sale and purchase of the Property, and thereafter at the risk of the Purchaser.

In the event of significant damage to the Property occurring before the time of closing by reason of fire, tempest, lightning, earthquake, flood or other Act of God, fire, explosion, riot, civil commotion, insurrection or war, then the Purchaser may, at its option:

- (a) cancel this Agreement in which case the Deposit shall be returned to the Purchaser; or
- (b) close the sale and require the Vendor to assign to the Purchaser the proceeds of any insurance claim relating to the Property.

13. CLOSING DOCUMENTS

The Vendor will convey the Property to the Purchaser by delivering to the Purchaser at or before the time of closing a freehold transfer (the "Transfer") in registerable form. The Purchaser shall bear the cost of the conveyance, and the Vendor shall bear the cost of clearing title of any charges other than Permitted Liens and shall be solely responsible for any real estate commission payable.

At least three business days prior to the day of closing the Purchaser shall cause its solicitors to prepare and deliver to the Vendor all documents reasonably required by the

Vendor's solicitors to complete this transaction in accordance with its terms including the Transfer, a statement of adjustments, an assignment of the Vendor's interest in any outstanding guarantees, warranties or indemnities with respect to the Property, an assignment and assumption of Permitted Liens, a mutual undertaking to readjust, the GST Certificate and any documents and certificates referred to herein and such other documents as may be reasonably necessary for more perfectly and absolutely transferring, assuring and vesting title to the Property in the Purchaser as contemplated hereby.

If the Vendor has existing financial charges to be cleared from title, as shown on Schedule B, the Vendor, while still required to clear such charges, may wait to pay and discharge existing financial charges until immediately after receipt of the Purchase Price, but in this event, the Purchaser shall pay the Purchase Price to a lawyer or notary in trust, on undertakings to pay and discharge the financial charges and remit the balance, if any to the Vendor.



If the Purchaser is relying upon a new mortgage to finance the Purchase Price the Purchaser, while still required to pay the Purchase Price on the Closing Date, may wait to pay the Purchase Price to the Vendor until after the transfer and new mortgage documents have been lodged for registration in the appropriate Land Title office, but only if, before such lodging, the Purchaser has: (a) made available for tender to the Vendor that portion of the Purchase Price not secured by the new mortgage, and (b) fulfilled all the new mortgagee's conditions for funding except lodging the mortgage for registration, and (c) made available to the Vendor, a lawyer's or notary's undertaking to pay the Purchase Price upon the lodging of the transfer and new mortgage documents and the advance by the mortgagee of the mortgage proceeds.

On or before the Closing Date the Purchaser will pay to the Purchaser's solicitors, in trust, the balance of the Purchase Price, as adjusted, less the amount to be advanced to the Purchaser on the Closing Date under any mortgage financing arranged by the Purchaser. Following such payment and after receipt by the Purchaser's solicitors of the documents and items referred to in this Clause 13, the Purchaser will cause the Purchaser's solicitors to file the Transfer (together with such other documents as are required to be filed) in the Land Title Office concurrently with any security documents applicable to any mortgage financing arranged by the Purchaser in connection with the purchase of the Property. Upon the Purchaser's solicitors obtaining a post application search of the Property which indicates that, in the normal routine of the Land Title Office, title to the Property will issue in the name of the Purchaser subject only to the Permitted Encumbrances and the state of title is in accordance with the terms and conditions of this Offer, the Purchaser shall cause the Purchaser's solicitors to pay to the Vendor or the Vendor's solicitors the balance of the Purchase Price and to release and deliver the closing documents referred to herein.

14. HEALTH CANADA LICENSES

Forthwith after the Closing Date, the Vendor, at its cost, will cause any Health Canada licenses associated with the Property to be terminated or extinguished.

15. SITE PROFILE

The Purchaser hereby waives any requirement for the Vendor to provide the Purchaser with a site profile for the Property under the *Environmental Management Act* of British Columbia or any regulation in respect thereto.

16. DISCLOSURE

The Vendor and the Purchaser acknowledge and agree that:

- (a) in accordance with the Code of Ethics of the Canadian Real Estate Association, Colliers Macaulay Nicolls Inc. (the "Purchaser's Agent" represented by Greg Lane (the "Salesperson") has disclosed that it is representing the Purchaser, and Macdonald Reality Westmar (the "Vendor's Agent" represented by Humraj Kallu, the "Salespeople") has disclosed that it is representing the Vendor in the transaction described in this Agreement;
- (b) the Agents, in order to accommodate the transaction described in this Agreement, were and are entitled to pass any relevant information they receive from either Party or from any other source to either of the Parties as the Agents see fit, without being in conflict of their duties to either Party; and
- (c) the Vendor will pay the Purchaser's Agent commission that will be due and payable on the Closing Date in the amount of Seventeen Thousand Dollars (\$17,000.00) plus GST.



17. CONFIDENTIALITY

The Parties agree that this Agreement, the terms contained herein and any negotiations in connection thereto shall be deemed "Confidential Information" and shall survive any termination or expiration of this Agreement.

18. BINDING AGREEMENT

Upon acceptance by both the Purchaser and the Vendor this Offer shall constitute a binding agreement for the purchase and sale of the Property on the terms and conditions contained herein.

19. EXECUTION BY ELECTRONIC MEANS

This Agreement may be executed by the parties in counterparts and transmitted by facsimile or electronically, and if so executed and transmitted, this Agreement will be, for all purposes, as effective as if the parties had delivered an executed original Agreement.

20. DATE OF ACCEPTANCE

This Offer is irrevocable and shall be open for acceptance by the Purchaser up to 5:00 p.m. PST on November 30, 2023 and, upon acceptance by the Purchaser, will constitute a binding agreement for the purchase and sale of the said Property on the terms and conditions contained herein. In the event that this Offer is not accepted by the Purchaser on or before the aforesaid time and date, then this Offer shall be null and void and the Refundable Deposit together with any accrued interest shall be returned to the Purchaser in full.

IN WITNESS WHEREOF the Vendor has executed this Offer the day and year first above written.

AVALITE SCIENCES INC.

Per: /s/ Punit Dhillon

Name: Punit Dhillon

Title: CEO

Per: /s/ Kait Arsenault

Name: Kait Arsenault

Title: CFO

V | P

THE PURCHASER HEREBY ACCEPTS the above Offer on the terms and conditions set out above, this 29th day of November, 2023.

TAB LABS INC.

Per:

/s/ John Stephen

Name: **John Stephen** Title: CEO

V P

SCHEDULE A

TITLE SEARCH PRINT 2023-06-08, 11:16:37

File Reference: Requester: Meredith Woods

CURRENT INFORMATION ONLY - NO CANCELLED INFORMATION SHOWN

Title Issued Under STRATA PROPERTY ACT (Section 249)

Land Title District NEW WESTMINSTER
Land Title Office NEW WESTMINSTER

Title Number CA7887379

From Title Number CA6961333

Application Received 2019-11-25
Application Entered 2019-12-09

Registered Owner in Fee Simple

Registered Owner/Mailing Address AVALITE SCIENCES, INC. INC.NO.BC0718177

104 - 9295 198TH STREET

LANGLEY, BC V1M 3J9

Taxation Authority Langley, The Corporation of the Township of

Description of Land

Parcel Identifier: 015-551-768

Legal Description:

STRATA LOT 4 SECTION 34 TOWNSHIP 8 NEW WESTMINSTER DISTRICT STRATA PLAN NW3095 TOGETHER WITH AN INTEREST IN THE COMMON PROPERTY IN PROPORTION TO THE UNIT ENTITLEMENT OF THE STRATA LOT AS SHOWN

ON FORM 1

Legal Notations NONE

Charges, Liens and Interests

Nature: STATUTORY BUILDING SCHEME

Registration Number AB186729

Registration Date and Time 1988-09-14 12:30

Remarks: INTER ALIA

LAND TITLE ACT SECTION 216

Duplicate Indefeasible Title NONE OUTSTANDING

Transfers NONE

Pending Applications NONE

Title Number: CA7887379 TITLE SEARCH PRINT Page 1 of 1

V P

Skye Bioscience and Tautomer Bioscience Enter Exclusive License for SBI-100 for Development and Sale of Products for Chronic Pain and Other Indications in South Africa and Rest of Africa

SAN DIEGO, CA, and JOHANNESBURG, SOUTH AFRICA, December 5, 2023 -- Skye Bioscience, Inc. (OTCQB: SKYE) ("Skye"),a pharmaceutical company developing drugs targeting the endocannabinoid system, and Tautomer Bioscience (Pty) Limited ("Tautomer"), a privately-owned biopharmaceutical company focused on addressing unmet medical needs in oncology, pain management and infectious diseases, announced today that they have entered into an agreement under which Skye is providing Tautomer with an exclusive license to develop and commercialize Skye's proprietary THC prodrug, SBI-100, as a novel suppository application for chronic intractable pain and other indications ("Tautomer Products") in South Africa and the rest of Africa.

Under the terms of the agreement Tautomer has the right to develop and commercialize products based on Skye's active pharmaceutical ingredient, SBI-100, formulated as a suppository, and related intellectual property. Tautomer is responsible for all formulation, preclinical and clinical development, drug product manufacturing and regulatory costs. Skye will receive development milestones, sales milestones, and tiered double-digit royalties on net product sales and will be compensated for all development work including the supply of SBI-100 active pharmaceutical ingredient ("API") for the Tautomer Products. Under the terms of the agreement Tautomer will also purchase Skye shares in open market transactions complying with Rule 10b-18 of the Securities Exchange Act of 1934, as amended. Skye has retained certain rights and options to obtain rights to the future use of new jointly developed intellectual property and other intellectual property owned or controlled by Tautomer related to SBI-100.

"Chronic pain causes disability, with significant personal and societal cost. According to PAIN SA, chronic pain affects 20% of the adult population in South Africa, with women and the elderly the worst affected. Our region, like the rest of the world, requires more non-opioid alternatives to address this growing epidemic," said Martin Magwaza, Managing Director of Tautomer. "South Africa has a progressive regulatory, legislative, and academic environment which embraces cannabinoids as therapies. Tautomer believes that the SBI-100 technology is an ideal candidate to harness the power of the endocannabinoid system to manage chronic pain as it provides predictable dosing and improved bioavailability, avoids first pass metabolism, and may potentially have safer outcomes."

There is growing evidence and interest in the role of the endocannabinoid system, in particular activation of the CB1 receptor ("CB1"), for the treatment of chronic pain. Multiple medical reviews have demonstrated that cannabis use can alleviate pain symptoms while potentially reducing or even substituting opioid and non-opioid pharmacologic pain treatments. Importantly, CB1 is highly expressed in areas of the brain and spinal cord associated with pain as well as in the dorsal root ganglia of the peripheral nervous system. With this widespread distribution of these receptors in such critical pain-processing regions, it has long been postulated that cannabinoids such as THC can directly modulate pain. However, the presence of bioavailability and safety issues, including THC's psychotropic side effects, associated with previously used methods of delivery have in the past considerably hampered the development of cannabinoids to treat chronic pain. There is a need for pharmaceutical dosage forms that deliver the therapeutic effects of THC and minimize adverse effects.

Skye's research has demonstrated that a THC prodrug delivered as a suppository offers significantly improved bioavailability, resulting in better pharmacokinetics and potentially improved safety. Tautomer believes that Skye's data is sufficient to support a regulatory filing in South Africa for the chronic pain indication.

"The persistent global challenge of chronic pain, despite the array of non-pharmacologic and pharmacologic treatment options, in our view creates a need for new analgesics with novel mechanisms of action. We commend the Tautomer team for their foresight in identifying the potential of SBI-100 to address this unmet medical need through an innovative formulation. Tautomer has a strong track record and a history of entrepreneurial success in medical product commercialization both in South Africa and across the broader African continent, and we are pleased to establish this agreement with them," said Punit Dhillon, CEO and Chair of Skye. "While Skye is developing its differentiated CB1-targeting pharmaceutical drugs for ophthalmic and cardiometabolic conditions, we also recognize the significant opportunities for our CB1 agonist in the realm of chronic pain and other medical indications. This a great opportunity to realize incremental value from our API using alternative drug delivery systems.

"We believe that Tautomer can pursue a potentially accelerated path to develop and obtain regulatory approval for new products with SBI-100 in South Africa in order to offer patients with chronic pain a non-opioid treatment alternative. We look forward to providing assistance to Tautomer to bring this product to market in a region with a significant unmet medical need," added Mr. Dhillon.

This license agreement is subject to approval by the Financial Surveillance Department of the South African Reserve Bank, as may be required in terms of the Exchange Control Regulations (issued in terms of the Currency and Exchanges Act 9 of 1933) for any of the transactions contemplated in this agreement.

Pain Market in South Africa

South African national surveys have reported a high prevalence of chronic pain in the adult population. Prevalence rates range from 14% to 37%. Women and the elderly are the worst affected. Chronic pain comes at a high cost to the individual as well as the health system and the economy at large. Furthermore, some reports have stated that as many as 50% of chronic pain sufferers do not achieve adequate pain relief, despite treatment. There is a need for more effective therapies for the management of chronic pain.

About SBI-100

SBI-100 is a proprietary amino acid ester prodrug of delta-9-tetrahydrocannabinol ("THC") that functions as an agonist, or activator, of the endocannabinoid system's CB1 receptor. This receptor plays an integral role in overall endocannabinoid system activities involved in modulating sleep, hunger/metabolism, pain, and other functions.

About Tautomer

Tautomer Bioscience is a privately owned, fully integrated health technology company based in Johannesburg, South Africa. Through our network of strategic partners, we engage in the development, manufacturing, and distribution of high-quality nuclear medicine-based therapies and diagnostics. Our primary focus in healthcare is to address the unmet medical needs in oncology, pain management, infectious diseases. For more information, please visit www.tautomer-bioscience.com.

About Skye Bioscience

Skye is focused on unlocking the pharmaceutical potential of the endocannabinoid system to treat diseases with inflammatory, fibrotic, and metabolic processes. Backed by leading life science venture investors, Skye's strategy leverages biologic targets with substantial human proof of mechanism for the development of first-in-class therapeutics with significant clinical and commercial differentiation. Nimacimab, a negative allosteric modulating antibody, inhibits peripheral CB1 with a favorable safety and tolerability profile in a Phase 1 study. Skye plans to start a Phase 2 cardiometabolic-focused study encompassing obesity for nimacimab in H1 2024. SBI-100 Ophthalmic Emulsion, a CB1 agonist, is currently being studied in a Phase 2

study with patients with glaucoma and ocular hypertension. For more information, please visit: https://www.skyebioscience.com.

CONTACT

Investor Relations

Email: ir@skyebioscience.com

Phone: (858) 410-0266

LifeSci Advisors, Mike Moyer

Email: mmoyer@lifesciadvisors.com

Phone: (617) 308-4306

FORWARD LOOKING STATEMENTS

This press release contains forward-looking statements, including statements regarding our product development, business strategy, timing of clinical trials and commercialization of cannabinoid-derived therapeutics. Such statements and other statements in this press release that are not descriptions of historical facts are forward-looking statements that are based on management's current expectations and assumptions and are subject to risks and uncertainties. If such risks or uncertainties materialize or such assumptions prove incorrect, our business, operating results, financial condition, and stock price could be materially negatively affected. In some cases, forward-looking statements can be identified by terminology including "anticipated," "plans," "goal," "focus," "aims," "intends," "believes," "can," "could," "challenge," "predictable," "will," "would," "may" or the negative of these terms or other comparable terminology. We operate in a rapidly changing environment and new risks emerge from time to time. As a result, it is not possible for our management to predict all risks, nor can we assess the impact of all factors on our business or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statements the Company may make. Risks and uncertainties that may cause actual results to differ materially include, among others, our capital resources, uncertainty regarding the results of future testing and development efforts and other risks that are described in the Risk Factors section of Skye's most recent annual or quarterly report filed with the Securities and Exchange Commission. Except as expressly required by law, Skye disclaims any intent or obligation to update these forward-looking statements.